ORIGINAL ARTICLE

# Quantized stabilization of strict-feedback nonlinear systems based on ISS cyclic-small-gain theorem

Tengfei Liu · Zhong-Ping Jiang · David J. Hill

Received: 29 April 2011 / Accepted: 16 February 2012 / Published online: 10 March 2012 © Springer-Verlag London Limited 2012

**Abstract** This paper proposes a new tool for quantized nonlinear control design of dynamic systems transformable into the dynamically perturbed strict-feedback form. To address the technical challenges arising from measurement and actuator quantization, a new approach based on set-valued maps is developed to transform the closed-loop quantized system into a large-scale system composed of input-to-state stable (ISS) subsystems. For each ISS subsystem, the inputs consist of quantization errors and interacting states, and moreover, the ISS gains can be assigned arbitrarily. Then, the recently developed cyclic-small-gain theorem is employed to guarantee input-to-state stability with respect to quantized system. Interestingly, it is shown that, under some realistic assumptions, any *n*-dimensional dynamically perturbed strict-feedback nonlinear system can be globally practically stabilized by a quantized control law using 2n three-level dynamic quantizers.

T. Liu (🖾) · Z.-P. Jiang Department of Electrical and Computer Engineering, Polytechnic Institute of New York University, Brooklyn, NY 11201, USA e-mail: tliu@poly.edu

Z.-P. Jiang e-mail: zjiang@poly.edu

This research was supported in part under the Australian Research Councils Discovery funding scheme (project number: FF0455875), in part by a seed grant from NYU and POLY, and by NSF grant DMS-0906659.

D. J. Hill School of Electrical and Information Engineering, The University of Sydney, Sydney, NSW 2006, Australia e-mail: djhill@sydney.edu.au

**Keywords** Quantized control · Input-to-state stability (ISS) · Small-gain theorem · Nonlinear systems · Dynamic quantization

# **1** Introduction

The quantized stabilization problem is motivated by the practical requirement that the signals of the control devices, including controller, actuators and sensors, are quantized before transmission through digital communication channels to other devices in the closed-loop control system. Here, a quantizer can be mathematically modeled as a discontinuous function from a continuous region to a discrete set of numbers. A practical quantizer has a finite number of quantization levels, and the quantized signal is saturated if the original signal is outside the range of the quantizer. This motivates the dynamic quantization [1], by which the quantization levels are appropriately scaled in the process of quantized control to achieve global stabilization with finite-level quantizers. In this paper, we propose a new design tool for quantized feedback stabilization of dynamically perturbed strict-feedback systems, an important class of nonlinear systems widely studied in the nonlinear control literature.

Quantization errors cause major problems for quantized stabilization; see, for instance, [1-5] for linear systems and [6-12] for nonlinear systems. In particular, the book [10] discussed the relation between space discretization and perturbation. The recent papers [4,6,7,12] characterize the quantization error as a sector bounded perturbation. Reference [6] studied the conditions under which the quantization error does not cancel the stabilizing effect of a dissipativity-based control law. Reference [7] achieved semi-global quantized stabilization for nonlinear systems based on the semi-global stabilization approach in [13]. In [14], the problem caused by the finitelevel quantizers was addressed for feedforward (upper-triangular) systems. Using the idea of scaling quantization levels, the authors of [1,9,11,15] studied dynamically quantized control with finite-level quantizers.

Quantized stabilization is closely related to robust stabilization by treating the state quantization errors as measurement errors. Despite its importance and relevance to many practical control problems, robust nonlinear stabilization with measurement errors has not been paid enough attention. In [16], it was discovered that if a system can be stabilized, then it can also be stabilized insensitively to small measurement errors. The authors of [17] introduced a robust nonlinear control design approach based on backstepping methodology (see e.g., [18]) and flattened Lyapunov functions to deal with bounded measurement errors. But with the method in [17], the influence of the measurement errors grows with the order of the system, which is daunting for high-order systems and for its further application to quantized stabilization.

The concept of input-to-state stability (ISS) invented by Sontag (see [19] for an excellent tutorial on ISS), appears to be fundamental to describe the robust stability of closed-loop quantized systems with respect to quantization errors in several results; see e.g., Liberzon and Nešić's dynamically quantized control results [9,11,20,21]. With the ISS small-gain theorem in [22] as a tool, reference [23] established a unified framework for control design of nonlinear systems with quantization and time scheduling via an emulation-like approach. Based on the small-gain theorem, the quantized

controller should usually be designed to make the control error caused by quantization error smaller than the range of the quantizer, to ascertain global stabilization with finite-level quantization [11,23].

The objective of this paper is to propose a design tool for global stabilization of strict-feedback nonlinear systems with both state quantization and actuator quantization, by means of the recent developments in cyclic-small-gain theorems (see, e.g., [24,25]). Our recent contributions to quantized control [12,26] demonstrate the usefulness of the cyclic-small-gain theorem. For other related network small-gain theorems, the reader should consult [22, 27-32]. In our novel controller design procedure, we will introduce a new set-valued map method to overcome the problems caused by the nonlinearity and discontinuity of the quantizers as well as the presence of dynamic uncertainty. Consequently, differential inclusions will be adopted to represent the closed-loop quantized system, and the motion of the closed-loop quantized system will be described with the extended Filippov solution introduced in [33,34]; see also [35] for the concept of Filippov solution. The gain assignment technique originally proposed in [22] will have to be further modified to adapt to the specifically designed set-valued maps such that the closed-loop quantized system can be transformed into a large-scale system composed of ISS subsystems. Moreover, the ISS gains of each subsystem will be assigned appropriately to satisfy the cyclic-small-gain condition and to guarantee the ISS of the closed-loop quantized system with respect to quantization errors. An ISS-Lyapunov function will also be constructed to evaluate the influence of the quantization error. Differently from the previously known results, the adjustment of the quantization levels will result in jumps of the state of the closed-loop quantized system. The dynamic quantization design method introduced in [11] will be modified for global quantized stabilization based on the ISS analysis of a specific discrete-time nonlinear system.

The rest of this paper is organized as follows. Section 2 presents some mathematical preliminaries. Section 3 describes the problem formulation and gives an expected quantized controller structure. In Sect. 4, we recursively design set-valued maps for the class of strict-feedback systems with quantized signals. Section 5 is devoted to the synthesis of dynamic quantization. Section 6 presents a remark on measurement feedback control as a by-product of the main result. Section 7 presents some conclusions and discusses some future directions. The proofs of some technical lemmas are in the Appendix.

#### 2 Mathematical preliminaries

To make the paper self-contained, recall that a function  $\gamma : \mathbb{R}_+ \to \mathbb{R}_+$  is positive definite if  $\gamma(s) > 0$  for all s > 0 and  $\gamma(0) = 0$ .  $\gamma : \mathbb{R}_+ \to \mathbb{R}_+$  is a  $\mathscr{K}$  function (denoted by  $\gamma \in \mathscr{K}$ ) if it is continuous, strictly increasing and  $\gamma(0) = 0$ ; it is a  $\mathscr{K}_{\infty}$ function (denoted by  $\gamma \in \mathscr{K}_{\infty}$ ) if it is a  $\mathscr{K}$  function and also satisfies  $\gamma(s) \to \infty$ as  $s \to \infty$ . A function  $\beta : \mathbb{R}_+ \times \mathbb{R}_+ \to \mathbb{R}_+$  is a  $\mathscr{K}\mathscr{L}$  (denoted by  $\beta \in \mathscr{K}\mathscr{L}$ ) if  $\beta(\cdot, t)$  is a  $\mathscr{K}$  function for each fixed t and  $\beta(s, t)$  is decreasing to zero as  $t \to \infty$  for each  $s \in \mathbb{R}_+$ . Id represents the identity function. For  $\gamma_1, \gamma_2 \in \mathscr{K}$ , inequality  $\gamma_1 < \gamma_2$ means  $\gamma_1(s) < \gamma_2(s)$  for all s > 0. The ISS cyclic-small-gain theorem proposed in [24,25,28] is a tool to analyze the ISS property of large-scale nonlinear systems in the following form:

$$\dot{x}_i = f_i(x, u_i), \quad i = 1, \dots, N$$
 (1)

where  $x_i \in \mathbb{R}^{n_i}$ ,  $x = [x_1^{\mathrm{T}}, \dots, x_N^{\mathrm{T}}]^{\mathrm{T}}$ ,  $u_i \in \mathbb{R}^{m_i}$  and  $f_i : \mathbb{R}^{n+m_i} \to \mathbb{R}^{n_i}$  with  $n = \sum_{i=1}^N n_i$  is a locally Lipschitz map.

Although smooth ISS-Lyapunov functions have usually been used in the literature, for large-scale system (1) with locally Lipschitz dynamics, the smoothness requirement on the ISS-Lyapunov function can be reduced to locally Lipschitz continuity [34,31]. According to Rademacher's theorem (see, e.g., [36, p. 216]), locally Lipschitz continuity implies continuous differentiability almost everywhere.

It is assumed that each  $x_i$ -subsystem (i = 1, ..., N) in (1) admits a locally Lipschitz ISS-Lyapunov function  $V_i : \mathbb{R}^{n_i} \to \mathbb{R}$  satisfying

1. There exist  $\underline{\alpha}_i, \overline{\alpha}_i \in \mathscr{K}_{\infty}$  such that

$$\underline{\alpha}_{i}(|x_{i}|) \leq V_{i}(x_{i}) \leq \overline{\alpha}_{i}(|x_{i}|), \quad \forall x_{i} \in \mathbb{R}^{n_{i}};$$
(2)

2. There exist  $\gamma_{ij} \in \mathscr{K} \cup \{0\} \ (j \neq i)$  and  $\gamma_i^u \in \mathscr{K} \cup \{0\}$  such that

$$V_{i}(x_{i}) \geq \max_{j=1,\dots,N; j\neq i} \left\{ \gamma_{ij}\left(V_{j}\left(x_{j}\right)\right), \gamma_{i}^{u}\left(|u_{i}|\right) \right\}$$
$$\Rightarrow \nabla V_{i}(x_{i}) f_{i}(x, u_{i}) \leq -\alpha_{i}\left(V_{i}\left(x_{i}\right)\right)$$
(3)

wherever  $\nabla V_i$  exists, where  $\alpha_i$  is continuous and positive definite.

The functions  $\gamma_{ij}$ ,  $\gamma_i^u$  are known as the ISS gains of the subsystems. The following theorem presents a cyclic-small-gain condition to guarantee the ISS property of the large-scale system (1) with state x and input  $u = [u_1^T, \dots, u_N^T]^T$ .

**Theorem 1** [24] Consider the large-scale nonlinear system (1). Assume each  $x_i$ -subsystem admits an ISS-Lyapunov function  $V_i$  satisfying (2) and (3). Then, the large-scale nonlinear system (1) is ISS if for r = 2, ..., N,

$$\gamma_{i_1i_2} \circ \gamma_{i_2i_3} \circ \dots \circ \gamma_{i_ri_1} < \mathrm{Id} \tag{4}$$

where  $1 \le i_k \le N$  and  $i_k \ne i_{k'}$  if  $k \ne k'$  for  $1 \le k \le r$ .

By considering the subsystems (1) as vertices and the gains as the weights of directed arcs (i.e., connections between subsystems), the interconnection structure of the large-scale nonlinear system can be represented with a system digraph. Condition (4) is called cyclic-small-gain condition and means that the composition of the ISS gains along every simple cycle in the large-scale nonlinear system is less than the identity function Id.

For the ISS gains  $\gamma_{ij}$ 's  $(1 \le i \le N, j \ne i)$  satisfying condition (4), according to Lemma A.1 of [27], we can find  $\mathscr{K}_{\infty}$  functions  $\hat{\gamma}_{ij}$ 's  $(1 \le i \le N, j \ne i)$  which are continuously differentiable on  $(0, \infty)$  and slightly larger than the corresponding

 $\gamma_{ij}$ 's such that condition (4) still holds by replacing the  $\gamma_{ij}$ 's with the  $\hat{\gamma}_{ij}$ 's. Motivated by the ISS-Lyapunov function construction in [27], a locally Lipschitz ISS-Lyapunov function can be constructed for the large-scale system (1) as

$$V(x) = \max_{i=1,...,n} \{\sigma_i(V_i(x_i))\}$$
(5)

where  $\sigma_i$ 's are specific compositions of the  $\hat{\gamma}_{(.)}$ 's.

The influence of the external input u can be represented as

$$\theta(u) = \max_{i=1,\dots,n} \{ \sigma_i \circ \gamma_i^u(|u_i|) \}.$$
(6)

Denote  $f(x, u) = [f_1^{\mathrm{T}}(x, u_1), \dots, f_N^{\mathrm{T}}(x, u_N)]^{\mathrm{T}}$ . With the Lyapunov-based ISS cyclic-small-gain theorem presented in [25], we have

$$V(x) \ge \theta(u) \Rightarrow \nabla V(x) f(x, u) \le -\alpha(V(x))$$
(7)

wherever  $\nabla V$  exists, with  $\alpha$  being a continuous and positive definite function.

The type of ISS-Lyapunov functions (5) was used by [31] for the construction of an ISS-Lyapunov function for the matrix-small-gain theorem. Also notice that (5) was utilized in [25] in the construction of ISS-Lyapunov function for the ISS cyclic-small-gain theorem.

Recently, the authors of [33,34] extended the concepts of ISS and ISS-Lyapunov function to discontinuous systems and also proposed an extended Filippov solution for interconnected discontinuous systems described by differential inclusions. Based on the concept of extended Filippov solution, a discontinuous version of the ISS small-gain theorem was developed. Motivated by the results in [33,34], a discontinuous version of the cyclic-small-gain theorem can also be developed for large-scale discontinuous systems:

$$\dot{x}_i \in F_i(x, u_i), \quad i = 1, \dots, N \tag{8}$$

where  $F_i : \mathbb{R}^{n+m_i} \rightsquigarrow \mathbb{R}^{n_i}$  is a convex, compact and upper semi-continuous set-valued map and the variables are defined in the same way as for (1).

References [33,34] considered the case of multiple Lyapunov functions to carry out a general ISS theory for discontinuous nonlinear systems. Here we just consider a simplified case where each  $x_i$ -subsystem in (8) admits a common Lyapunov function  $V_i$  satisfying (2) and

$$V_{i}(x_{i}) \geq \max_{\substack{j=1,\dots,N; \ j\neq i}} \{\gamma_{ij}(V_{j}(x_{j})), \gamma_{i}^{u}(|u_{i}|)\}$$
  
$$\Rightarrow \max_{f_{i}\in F_{i}(x,u_{i})} \nabla V_{i}(x_{i}) f_{i} \leq -\alpha_{i}(V_{i}(x_{i}))$$
(9)

wherever  $\nabla V_i$  exists, which is a modification of (3).

With the cyclic-small-gain condition (4) satisfied by the large-scale discontinuous system, an ISS-Lyapunov function V can be constructed in a quite similar way as in (5), and property (7) should be modified as

$$V(x) \ge \theta(u) \Rightarrow \max_{f \in F(x,u)} \nabla V(x) f \le -\alpha(V(x))$$
(10)

wherever  $\nabla V$  exists, with  $F(x, u) = [F_1^{\mathrm{T}}(x, u_1), \dots, F_N^{\mathrm{T}}(x, u_N)]^{\mathrm{T}}$ .

As we will show in this paper, the above discontinuous variant plays a central role in the recursive quantized control design for dynamic networks or high-dimensional nonlinear systems transformable into a dynamic network.

## **3** Problem formulation

#### 3.1 System description

In this paper, we study quantized control of the following dynamically perturbed strict-feedback nonlinear system:

$$\dot{z} = g(z, x_1) \tag{11}$$

$$\dot{x}_i = x_{i+1} + \Delta_i(\bar{x}_i, z), \quad i = 1, \dots, n$$
 (12)

$$x_{n+1} \stackrel{\text{def}}{=} u \tag{13}$$

where  $[x_1, \ldots, x_n]^T := x \in \mathbb{R}^n$  is the measurable state,  $z \in \mathbb{R}^{n_z}$  represents the state of the inverse dynamics and is not measurable,  $u \in \mathbb{R}$  is the control input,  $\bar{x}_i = [x_1, \ldots, x_i]^T$ , and  $\Delta_i$ 's  $(i = 1, \ldots, n)$  are unknown locally Lipschitz continuous functions. We study the general case in which both the measurement x and the control input u are quantized.

Throughout the paper, the following assumptions are made on system (11)-(13).

**Assumption 1** System (11)–(13) with u = 0 is forward complete and small-time final-state norm-observable with x as the output, i.e., for u = 0,

$$\forall t_d > 0 \ \exists \varphi \in \mathscr{K}_{\infty} \text{ such that} |X(t_d)| \le \varphi(\|x\|_{[0,t_d]}), \quad \forall X(0) \in \mathbb{R}^{n+n_z}$$
 (14)

where  $X := [z^{T}, x^{T}]^{T}$ .

*Remark 1* Reference [37] discussed the equivalent characterizations of initial-state norm-observability, final-state norm-observability and  $\mathcal{KL}$  norm-observability for forward complete (or unboundedness observable) nonlinear systems with external inputs. In Assumption 1, we just use the concept of small-time final-state norm-observability for nonlinear systems with no external inputs.

*Remark 2* Assumption 1 is needed for global quantized stabilization; see [21]. However, it is important to mention that Assumption 1 is not needed if a bound is known on the initial state of system (11)–(13). In this case, only semi-global quantized stabilization can be achieved. Also, see [15,23].

**Assumption 2** For each  $\Delta_i$  with i = 1, ..., n, there exists a known  $\lambda_{\Delta_i} \in \mathscr{K}_{\infty}$  such that for all  $\bar{x}_i, z$ ,

$$|\Delta_i(\bar{x}_i, z)| \le \lambda_{\Delta_i}(|(\bar{x}_i, z)|).$$
(15)

Assumption 3 The *z*-subsystem (11) with  $x_1$  as the input admits a locally Lipschitz ISS-Lyapunov function  $V_0 : \mathbb{R}^{n_z} \to \mathbb{R}_+$  satisfying

1. There exist  $\underline{\alpha}_0, \overline{\alpha}_0 \in \mathscr{K}_\infty$  such that

$$\underline{\alpha}_{0}(|z|) \leq V_{0}(z) \leq \overline{\alpha}_{0}(|z|), \quad \forall z \in \mathbb{R}^{n_{z}};$$
(16)

2. There exist a  $\chi_z^{x_1} \in \mathscr{K}$  and a continuous and positive definite  $\alpha_0$  such that

$$V_0(z) \ge \chi_z^{x_1}(|x_1|) \Rightarrow \nabla V_0(z)g(z, x_1) \le -\alpha_0(V_0(z))$$
(17)

wherever  $\nabla V_0$  exists.

*Remark 3* Under Assumption 3, system (11)–(13) represents an important class of minimum-phase nonlinear systems, which have been studied extensively by many authors in the context of (non-quantized) robust and adaptive nonlinear control. The reader may consult [13,18,38,39] and references therein for the details.

#### 3.2 Quantization

In this paper, a quantizer  $q(r, \mu)$  with original signal  $r \in \mathbb{R}$  and variable  $\mu > 0$ is defined as  $q(r, \mu) = \mu q^o(r/\mu)$ , where  $q^o : \mathbb{R} \to \mathbb{R}$  is a piecewise constant function for which there exists a constant M > 0 such that  $|q^o(a) - M| \le 1$  if a > M;  $|q^o(a) - a| \le 1$  if  $|a| \le M$ ;  $|q^o(a) + M| \le 1$  if a < -M;  $q^o(0) = 0$ .

Then, quantizer  $q(r, \mu)$  satisfies:

I

$$|q(r,\mu) - M\mu| \le \mu, \quad \text{if } r > M\mu; \tag{18}$$

$$q(r,\mu) - r| \le \mu, \qquad \text{if } |r| \le M\mu; \tag{19}$$

$$|q(r,\mu) + M\mu| \le \mu, \text{ if } r < -M\mu;$$
 (20)

$$q(0,\mu) = 0. (21)$$

 $M\mu$  is the quantization range of quantizer  $q(r, \mu)$  and  $\mu$  represents the quantization error bound (i.e., the maximum of  $|q(r, \mu) - r|$  when  $|r| \le M\mu$ ). In dynamic quantization design,  $\mu$  is known as the "zooming" variable. Given fixed M, the basic idea of dynamic quantization is to dynamically update  $\mu$  (and thus  $M\mu$ ) in discrete-time to improve the control performance in the existence of the quantization error. Increasing  $\mu$ , referred to as "zooming-out", enlarges the quantization error bound  $\mu$  and the quantization range  $M\mu$ . Decreasing  $\mu$ , referred to as "zooming-in", reduces the quantization error bound  $\mu$  and the quantization range  $M\mu$ . *Remark 4* In several previously known quantized control results (see e.g., [15]), two positive parameters, say M',  $\delta'$ , are used to formulate a quantizer q' as:  $|q'(r, \mu') - r| \le \delta'\mu'$  if  $|r| \le M'\mu'$ ;  $|q'(r, \mu')| > (M' - \delta')\mu'$  if  $|r| > M'\mu'$ . Indeed, if we define  $M = M'/\delta'$ ,  $\mu = \delta'\mu'$  and a new quantizer  $q(r, \mu) = q'(r, \mu/\delta')$ , then we can see properties (18)–(20) hold for the new quantizer q. Moreover, properties (18) and (20) explicitly represent the saturation property of the quantizer. It will be demonstrated that the explicit description of saturation in (18) and (20) is necessary in our recursive control design later, in which we try to take the saturation property into account.

*Remark 5* It is shown in [40] that the simplest three-level quantizer satisfies the properties (18)–(20) with M = 3, and a larger number of levels means a larger M.

## 3.3 Quantized controller structure and control objective

In this paper, we introduce a new quantized control structure following an ISS smallgain design approach. Based on the ISS small-gain design method proposed in [41], we can recursively design a non-quantized controller for system (11)–(13) as

$$v_i = \breve{\kappa}_i (x_i - v_{i-1}), \quad i = 1, \dots, n-1$$
 (22)

$$u = \breve{\kappa}_n (x_n - v_{n-1}) \tag{23}$$

where  $v_0 = 0$  and  $\check{\kappa}_i$ 's are appropriately designed continuous functions for i = 1, ..., n. The maps defined in (22) are usually known as virtual control laws.

Our solution to the quantized control problem of system (11)–(13) is to add quantizers before and after each (virtual) control law defined in (22) through a recursive design approach.

Following this idea, the quantized controller would be in the following form:

$$v_i = q_{i2}(\kappa_i(q_{i1}(x_i - v_{i-1}, \mu_{i1})), \mu_{i2}), \quad i = 1, \dots, n-1$$
(24)

$$u = q_{n2}(\kappa_n(q_{n1}(x_n - v_{n-1}, \mu_{n1})), \mu_{n2})$$
(25)

where  $v_0 = 0$ , the  $q_{ij}$ 's are quantizers with zooming variables  $\mu_{ij}$ 's for i = 1, ..., n, j = 1, 2, and the  $\kappa_i$ 's for i = 1, ..., n are nonlinear functions. The  $\kappa_i$  in (24)–(25) is not necessarily the same as the  $\breve{\kappa}_i$  in (22)–(23) because of the implementation of the quantizers.

*Remark 6* The new quantized controller structure (24)–(25) leads to no restrictiveness in practical applications. Indeed, the quantized control structure making use of adjustable centers of the quantizers can also be found in the very recent work [42].

We make the following assumption on all the employed quantizers in (24)–(25).

**Assumption 4** Each quantizer  $q_{ij}$  for i = 1, ..., n, j = 1, 2 with zooming variable  $\mu_{ij}$  satisfies

$$|q_{ij}(r,\mu_{ij}) - M_{ij}\mu_{ij}| \le \mu_{ij}, \quad \text{if } r > M_{ij}\mu_{ij}; \tag{26}$$

$$|q_{ij}(r,\mu_{ij}) - r| \le \mu_{ij}, \qquad \text{if } |r| \le M_{ij}\mu_{ij};$$
 (27)

$$|q_{ij}(r,\mu_{ij}) + M_{ij}\mu_{ij}| \le \mu_{ij}, \quad \text{if } r < -M_{ij}\mu_{ij}; \tag{28}$$

$$q_{ij}(0,\mu_{ij}) = 0. (29)$$

where  $M_{i1} > 2$  and  $M_{i2} > 1$ .

In dynamic quantization, each zooming variable  $\mu_{ij} > 0$  for i = 1, ..., n, j = 1, 2is a piecewise constant signal and updated in discrete-time. Without loss of generality, we assume that the piecewise constant zooming variables are right-continuous on the time-line. Motivated by Liberzon [15], we will still design a dynamic quantization composed of two stages, zooming-out stage and zooming-in stage. In this paper, we consider the case where the update time sequences for all the zooming variables are the same and denoted by  $\{t_0, t_1, t_2, ...\}$ , in which  $t_{k+1} - t_k = t_d$  with constant  $t_d > 0$ for  $k \in \mathbb{Z}_+$ .

The update dynamics of each  $\mu_{ij}$  is expected to be in the following form:

$$\mu_{ij}(t_{k+1}) = Q_{ij}(\mu_{ij}(t_k)), \quad k \in \mathbb{Z}_+.$$
(30)

In the zooming-out stage,  $Q_{ij} = Q_{ij}^{out}$ ; in the zooming-in stage,  $Q_{ij} = Q_{ij}^{in}$ .

The objective of this paper is to design a quantized controller in the form of (24)–(25) with dynamic quantization (30) to globally stabilize system (11)–(13) such that all the signals including the state *x* in the closed-loop quantized system are bounded, and moreover, to steer  $x_1$  to an arbitrarily small neighborhood of the origin.

The configuration of the total quantized control system is shown in Fig. 1.

#### 4 Recursive control design with set-valued maps-static quantization

A technical obstacle for quantized feedback control design is that the quantized control system in question must be made robust with respect to the quantization errors. The nonlinearity and dimensionality of system (11)–(13) and the saturation and discontinuity of quantization cause the major difficulties.

The objective of this section is to develop a recursive design procedure for  $\kappa_i$  in (24)–(25) by taking into account the effects of static quantization, such that the closed-loop quantized system admits nested invariant sets for further dynamic quantization designs. In Sect. 4.1, we introduce set-valued maps  $S_i$  depending on  $\kappa_i$   $(1 \le i \le n)$  to a recursive design to deal with the quantization errors and the closed-loop system will be transformed into a large-scale system composed of  $e_i$ -subsystems represented by differential inclusions. Section 4.2 presents a modified gain assignment technique to render the  $e_i$ -subsystems ISS with *any* specified ISS gains by appropriately choosing the set-valued maps. In Sect. 4.3, we show that we can find quantized control laws



Fig. 1 The quantized control structure

in the form of (24)–(25) belonging to the chosen set-valued maps. The cyclic-smallgain theorem is employed to guarantee the ISS property and the existence of nested invariant sets for the closed-loop quantized system in Sect. 4.4.

In this section, we make the following assumption on the zooming variables.

**Assumption 5** For i = 1, ..., n, j = 1, 2, the zooming variable  $\mu_{ij}$  is constant with respect to time.

## 4.1 Main steps of the recursive design

## 4.1.1 Initial step: the e<sub>1</sub>-subsystem

Let  $e_1 = x_1$ . The  $e_1$ -subsystem is in the following form:

$$\dot{e}_1 = x_2 + \Delta_1(\bar{x}_1, z). \tag{31}$$

Define a set-valued map  $S_1$  as

$$S_1(\bar{x}_1, \mu_{11}, \mu_{12}) = \{\kappa_1(x_1 + b_{11}) + b_{12} : |b_{11}| \le \max\{c_{11}|e_1|, \mu_{11}\}, |b_{12}| \le \mu_{12}\}$$
(32)

where  $\kappa_1$  is a continuously differentiable, odd, strictly decreasing and radially unbounded function and  $0 < c_{11} < 1$  is a constant, both of which will be determined later. It should be noted that  $b_{11}$ ,  $b_{12}$  defined in (32) are used as auxiliary variables to define set-valued map  $S_1$ .

Define

$$e_2 = \mathbf{d}(x_2, S_1(\bar{x}_1, \mu_{11}, \mu_{12})) \tag{33}$$

where **d** represents the directed distance from a point to an interval and is defined as

$$\mathbf{d}(\xi, \Omega) = \begin{cases} \xi - \max \Omega, & \text{if } \xi > \max \Omega; \\ \xi - \min \Omega, & \text{if } \xi < \min \Omega; \\ 0, & \text{otherwise.} \end{cases}$$
(34)

where  $\xi \in \mathbb{R}$  and  $\Omega \subset \mathbb{R}$  is an interval.

We rewrite the  $e_1$ -subsystem (31) as

$$\dot{e}_1 = x_2 - e_2 + \Delta_1(\bar{x}_1, z) + e_2 \tag{35}$$

where we have  $x_2 - e_2 \in S_1(\bar{x}_1, \mu_{11}, \mu_{12})$  from (34).

*Remark* 7 It is necessary to give a more detailed description of the set-valued map  $S_1$ . Consider the first-order nonlinear system  $\dot{e}_1 = x_2 + \Delta_1(\bar{x}_1, z)$ . Note that  $e_1 = x_1$ . With the gain assignment technique in [22,41,43], we can design a control law  $x_2 = \kappa_1(x_1)$  to stabilize the  $e_1$ -system. In the existence of quantization errors, control law  $x_2 = \kappa_1(x_1)$  should be modified as  $x_2 = q_{12}(\kappa_1(q_{11}(x_1, \mu_{11})), \mu_{12})$ . Clearly, the set-valued map  $S_1$  takes into account the quantization errors of both quantizers  $q_{11}$  and  $q_{12}$ . In the control design procedure below, we will recursively design new set-valued maps and define new subsystems based on a similar idea.

*Remark 8* In our recent paper [12], we employed set-valued maps to cover the sector-bounded uncertainties caused by logarithmic quantizers. In [12], the saturation property of the quantizers was not taken into account and the quantization levels are fixed. Moreover, in [12], the quantizers are directly connected to the state measurements, which is different from the quantized control structure (24)–(25).

#### 4.1.2 Recursive step: the $e_i$ -subsystems

Denote  $\bar{\mu}_{i1} = [\mu_{11}, \dots, \mu_{i1}]^{\mathrm{T}}$  and  $\bar{\mu}_{i2} = [\mu_{12}, \dots, \mu_{i2}]^{\mathrm{T}}$  for  $i = 1, \dots, n$ . For each  $i = 2, \dots, n$ , define a set-valued map  $S_i$  as

$$S_{i}(\bar{x}_{i}, \bar{\mu}_{i1}, \bar{\mu}_{i2}) = \{\kappa_{i}(x_{i} - \varsigma_{i-1} + b_{i1}) + b_{i2} : \varsigma_{i-1} \in S_{i-1}(\bar{x}_{i-1}, \bar{\mu}_{(i-1)1}, \bar{\mu}_{(i-1)2}), \\ |b_{i1}| \le \max\{c_{i1}|e_{i}|, \mu_{i1}\}, |b_{i2}| \le \mu_{i2}\}$$
(36)

where  $\kappa_i$  is a continuously differentiable, odd, strictly decreasing and radially unbounded function and  $0 < c_{i1} < 1$  is a constant, both of which will be determined later. The definition of  $S_i$  guarantees its convexity, compactness and upper semicontinuity of the set-valued map  $S_i$ . Here,  $b_{i1}, b_{i2}$  are auxiliary variables used to define the set-valued map  $S_i$ . *Remark* 9 It can be observed that  $S_1(\bar{x}_1, \bar{\mu}_{11}, \bar{\mu}_{12})$  defined in (32) is in the form of (36) with  $S_0(\bar{x}_0, \bar{\mu}_{01}, \bar{\mu}_{02}) := \{0\}.$ 

For each  $i = 2, \ldots, n$ , define  $e_{i+1}$  as

$$e_{i+1} = \mathbf{d}(x_{i+1}, S_i(\bar{x}_i, \bar{\mu}_{i1}, \bar{\mu}_{i2})).$$
(37)

Lemma 1 shows that with the recursive construction of set-valued maps and new state variables in (36) and (37), we can represent the  $e_i$ -subsystems for i = 1, ..., n with differential inclusions.

**Lemma 1** Consider the  $(x_1, ..., x_n)$ -system in (12)–(13). Under Assumptions 2 and 5, with the definitions in (32), (33), (36) and (37), each  $e_i$ -subsystem for  $1 \le i \le n$  can be represented with the differential inclusion:

$$\dot{e}_i \in S_i(\bar{x}_i, \bar{\mu}_{i1}, \bar{\mu}_{i2}) + \Phi_i^*(e_{i+1}, \bar{x}_i, \bar{\mu}_{(i-1)1}, \bar{\mu}_{(i-1)2}, z)$$
 (38)

where  $\Phi_i^*$  is a convex, compact and upper semi-continuous set-valued map, and there exists a  $\lambda_{\Phi_i^*} \in \mathscr{K}_{\infty}$  such that for all  $(e_{i+1}, \bar{x}_i, \bar{\mu}_{(i-1)1}, \bar{\mu}_{(i-1)2}, z)$ , any  $\phi_i^* \in \Phi_i^*(e_{i+1}, \bar{x}_i, \bar{\mu}_{(i-1)1}, \bar{\mu}_{(i-1)2}, z)$  satisfies

$$|\phi_i^*| \le \lambda_{\Phi_i^*}(|(\bar{e}_{i+1}, \bar{\mu}_{(i-1)1}, \bar{\mu}_{(i-1)2}, z)|)$$
(39)

where  $\bar{e}_i := [e_1, \ldots, e_i]^{\mathrm{T}}$ .

The proof of Lemma 1 is in Appendix A.

With Lemma 1, through the recursive design approach, we have transformed the  $(x_1, \ldots, x_n)$ -system into the new  $(e_1, \ldots, e_n)$ -system with each  $e_i$ -subsystem  $(i = 1, \ldots, n)$  of form (38). The extended Filippov solution of each  $e_i$ -subsystem can be defined with differential inclusion (38) because both set-valued maps  $S_i$  and  $\Phi_i^*$  are convex, compact and upper semi-continuous; see [33,34].

## 4.2 ISS of the subsystems

For each  $e_i$ -subsystem with i = 1, ..., n, we define the following ISS-Lyapunov function candidate

$$V_i(e_i) = \alpha_V(|e_i|) \tag{40}$$

where  $\alpha_V(s) = \frac{1}{2}s^2$  for  $s \in \mathbb{R}_+$ . For convenience of notations, define  $V_{n+1}(e_{n+1}) = \alpha_V(|e_{n+1}|)$ . We simply use  $V_i$  instead of  $V_i(e_i)$  in the following discussions.

Denote  $e_0 = z$ . Then, we have

$$\dot{e}_0 = g(e_0, e_1).$$
 (41)

Define  $\gamma_{e_0}^{e_1}(s) = \chi_z^{x_1} \circ \alpha_V^{-1}(s)$  for  $s \in \mathbb{R}_+$ . With Assumption 3 satisfied, we have

$$V_0 \ge \gamma_{e_0}^{e_1}(V_1) \Rightarrow \nabla V_0 g(e_0, e_1) \le -\alpha_0(V_0).$$
(42)

Lemma 2 is a modification of the gain assignment technique employed in the previous ISS small-gain design results [22,41,43], and states that, for i = 1, ..., n, by appropriately designing  $\kappa_i$ , each  $e_i$ -subsystem can be rendered to be ISS with  $V_i$ defined in (40) as ISS-Lyapunov function.

**Lemma 2** Consider the  $e_i$ -subsystem (i = 1, ..., n) in the form of (38) with  $S_i$ defined in (32) and (36). Under Assumptions 2 and 5, for any specified constants  $\varepsilon_i > 0, \iota_i > 0, 0 < c_{i1}, c_{i2} < 1, \gamma_{e_i}^{e_k} \in \mathscr{K}_{\infty}$  for k = 0, ..., i - 1, i + 1, and  $\gamma_{e_i}^{\mu_{k1}}, \gamma_{e_i}^{\mu_{k2}} \in \mathscr{K}_{\infty}$  for k = 1, ..., i - 1, one can find a continuously differentiable, odd, strictly decreasing and radially unbounded  $\kappa_i$  for the set-valued map  $S_i$  such that the  $e_i$ -subsystem is ISS with  $V_i(e_i) = \alpha_V(|e_i|) = \frac{1}{2}e_i^2$  as an ISS-Lyapunov function satisfying

$$V_{i} \geq \max_{k=1,...,i-1} \left\{ \begin{array}{l} \gamma_{e_{i}}^{e_{0}}(V_{0}), \gamma_{e_{i}}^{e_{k}}(V_{k}), \gamma_{e_{i}}^{e_{i+1}}(V_{i+1}), \\ \gamma_{e_{i}}^{\mu_{k1}}(\mu_{k1}), \gamma_{e_{i}}^{\mu_{k2}}(\mu_{k2}), \gamma_{e_{i}}^{\mu_{i1}}(\mu_{i1}), \gamma_{e_{i}}^{\mu_{i2}}(\mu_{i2}), \varepsilon_{i} \end{array} \right\}$$
  
$$\Rightarrow \max_{\psi_{i} \in \Psi_{i}(e_{i+1}, \bar{x}_{i}, \bar{\mu}_{i1}, \bar{\mu}_{i2}, z)} \nabla V_{i} \psi_{i} \leq -\iota_{i} V_{i}(e_{i})$$
(43)

where

$$\gamma_{e_i}^{\mu_{i1}}(s) = \alpha_V \left(\frac{1}{c_{i1}}s\right) \tag{44}$$

$$\gamma_{e_i}^{\mu_{i2}}(s) = \alpha_V \left( \frac{1}{1 - c_{i1}} \bar{\kappa}_i^{-1} \left( \frac{1}{c_{i2}} s \right) \right)$$
(45)

$$\Psi_i(e_{i+1}, \bar{x}_i, \bar{\mu}_{i1}, \bar{\mu}_{i2}, z) := S_i(\bar{x}_i, \bar{\mu}_{i1}, \bar{\mu}_{i2}) + \Phi_i^*(e_{i+1}, \bar{x}_i, \bar{\mu}_{(i-1)1}, \bar{\mu}_{(i-1)2}, z)$$
(46)

with  $\bar{\kappa}_i(s) = |\kappa_i(s)|$  for  $s \in \mathbb{R}_+$ .

The proof of Lemma 2 is in Appendix **B**.

*Remark 10* The gain assignment lemmas in the earlier papers [22,41,43] deal with nonlinear systems represented with differential equations but not differential inclusions. Furthermore, in our control design for the  $e_i$ -subsystem, we deal with the effects of the set-valued map  $S_{i-1}$ , the control error  $c_{i1}|e_i|$  and the quantization errors  $\mu_{i1}$  and  $\mu_{i2}$ . Also, it is worth noting that (43) yields a *common* Lyapunov function for each  $e_i$ -subsystem, while the general result on ISS discontinuous systems in [33,34] relies upon multiple Lyapunov functions.

## 4.3 Quantized controller

In the subsections above, we designed set-valued maps to derive a large-scale system composed of ISS subsystems described by differential inclusions. In this subsection,

we show that the quantized control law u in the form of (24)–(25) with the  $\kappa_i$ 's defined above belongs to the set-valued map  $S_n$  under realizable conditions. In this way, the closed-loop quantized system with the designed quantized control law u can be represented with a large-scale system composed of ISS subsystems.

Recall  $\bar{\kappa}_i(s) = |\kappa_i(s)|$  for  $s \in \mathbb{R}_+$ . Lemma 3 provides conditions under which the quantized control law *u* in the form of (24)–(25) belongs to the set-valued map  $S_n$ .

Lemma 3 Under Assumption 4, if

$$\frac{1}{M_{i1}} < c_{i1} \le 0.5, \quad \frac{1}{M_{i2}} < c_{i2} < 1 \tag{47}$$

for all  $i = 1, \ldots, n$  and if

$$|e_i| \le M_{i1}\mu_{i1},\tag{48}$$

$$\bar{\kappa}_i((1 - c_{i1})|e_i|) \le M_{i2}\mu_{i2} \tag{49}$$

for all i = 1, ..., n, then  $v_i$  for i = 1, ..., n - 1 and u defined in (24)–(25) satisfy

$$v_i \in S_i(\bar{x}_i, \bar{\mu}_{i1}, \bar{\mu}_{i2}), \quad i = 1, \dots, n-1,$$
(50)

$$u \in S_n(\bar{x}_n, \bar{\mu}_{n1}, \bar{\mu}_{n2}).$$
 (51)

The proof of Lemma 3 is given in Appendix C by fully using the properties of the quantizers and the set-valued maps.

*Remark 11* Conditions (48) and (49) imply that the signals  $|e_i|$  and  $|\bar{k}_i((1 - c_i)|e_i|)|$  should be covered by the quantization ranges  $M_{i1}\mu_{i1}$  and  $M_{i2}\mu_{i2}$ , respectively, such that the quantized control law *u* belongs to the set-valued map  $S_n$ . This is caused by the saturation property of the quantizers.

# 4.4 Small-gain based synthesis and nested invariant sets of the closed-loop quantized system

Recall  $e = [e_0^T, e_1, \ldots, e_n]^T$ . By considering the  $e_i$ -subsystems as vertices and the gain interconnections between them as directed arcs, the interconnection structure of the  $(e_0, e_1, \ldots, e_n)$ -system can be represented with a digraph, as shown in Fig. 2. The purpose of this subsection is to fine tune the ISS gains to yield ISS property of the closed-loop quantized system with e as the state by using the cyclic-small-gain theorem.

**Fig. 2** The interconnection digraph of the  $(e_0, e_1, \ldots, e_n)$ -system



Recall  $\bar{e}_i = [e_1, \ldots, e_i]^T$ . For each  $(e_0, \bar{e}_i)$ -subsystem  $(i = 1, \ldots, n)$ , given the  $(e_0, \bar{e}_{i-1})$ -subsystem, by designing the set-valued map  $S_i$  for the  $e_i$ -subsystem, we can assign the ISS gains from states  $e_0, \ldots, e_{i-1}$  to state  $e_i$ . According to the recursive design, we assign the ISS gains  $\gamma_{e_i}^{e_k}$  for  $k = 0, \ldots, i - 1$  such that

$$\gamma_{e_{0}}^{e_{1}} \circ \gamma_{e_{1}}^{e_{2}} \circ \gamma_{e_{2}}^{e_{3}} \circ \cdots \circ \gamma_{e_{i-1}}^{e_{i}} \circ \gamma_{e_{i}}^{e_{0}} < \mathrm{Id}$$

$$\gamma_{e_{1}}^{e_{2}} \circ \gamma_{e_{2}}^{e_{3}} \circ \cdots \circ \gamma_{e_{i-1}}^{e_{i}} \circ \gamma_{e_{i}}^{e_{1}} < \mathrm{Id}$$

$$\vdots$$

$$\gamma_{e_{i-1}}^{e_{i}} \circ \gamma_{e_{i}}^{e_{i-1}} < \mathrm{Id}$$
(52)

where Id represents the identity function. Applying this reasoning repeatedly, we can guarantee (52) for all i = 1, ..., n. In this way, the *e*-system satisfies the cyclic-small-gain condition.

In the digraph of the *e*-system shown in Fig. 2, the  $e_1$ -subsystem is reachable from the subsystems of  $e_0, e_2, \ldots, e_n$ , i.e., there are sequences of directed arcs from the subsystems of  $e_0, e_2, \ldots, e_n$  to the  $e_1$ -subsystem.

Motivated by the ISS-Lyapunov function construction in [27], we construct an ISS-Lyapunov function candidate for the *e*-system as

$$V(e) = \max_{i=0,...,n} \{\sigma_i(V_i(e_i))\}$$
(53)

with  $\sigma_1(s) = s$ ,  $\sigma_i(s) = \hat{\gamma}_{e_1}^{e_2} \circ \cdots \circ \hat{\gamma}_{e_{i-1}}^{e_i}(s)$  (i = 2, ..., n) and  $\sigma_0(s) = \max_{i=1,...,n} \{\sigma_i \circ \hat{\gamma}_{e_i}^{e_0}(s)\}$  for  $s \in \mathbb{R}_+$ , where the  $\hat{\gamma}_{(\cdot)}^{(\cdot)}$ 's are  $\mathscr{K}_{\infty}$  functions continuously differentiable on  $(0, \infty)$  and slightly larger than the corresponding  $\gamma_{(\cdot)}^{(\cdot)}$ 's, and still satisfy the cyclic-small-gain condition.

The following lemma states that we can appropriately choose the  $\kappa_i$ 's for the setvalued maps  $S_i$ 's such that the cyclic-small-gain condition (52) is satisfied and the closed-loop quantized system with state *e* admits specific ISS properties.

**Lemma 4** Consider the e-system composed of the  $e_i$ -subsystems in the form of (41) and (38) satisfying (42) and (43). If the ISS gains defined in (42) and (43) satisfy (52) for all i = 1, ..., n and if  $u \in S_n(\bar{x}_n, \bar{\mu}_{n1}, \bar{\mu}_{n2})$ , then the ISS-Lyapunov function candidate V defined in (53) for the e-system satisfies

$$V(e) \ge \theta(\bar{\mu}_{n1}, \bar{\mu}_{n2}, \bar{e}_n) \Rightarrow \max_{\psi \in \Psi(e, x, \bar{\mu}_{n1}, \bar{\mu}_{n2})} \nabla V(e) \psi \le -\alpha(V(e))$$
(54)

wherever  $\nabla V$  exists, where  $\alpha$  is a continuous and positive definite function, and

$$\theta(\bar{\mu}_{n1}, \bar{\mu}_{n2}, \bar{\varepsilon}_n) := \max_{i=1,\dots,n} \left\{ \sigma_i \left( \max_{k=1,\dots,i} \{ \gamma_{e_i}^{\mu_{k1}}(\mu_{k1}), \gamma_{e_i}^{\mu_{k2}}(\mu_{k2}), \varepsilon_i \} \right) \right\}$$
(55)

$$\Psi(e, x, \bar{\mu}_{n1}, \bar{\mu}_{n2}) := [\{g^{\mathrm{T}}(e_0, e_1)\}, \Psi_1(e_2, \bar{x}_1, \bar{\mu}_{11}, \bar{\mu}_{12}), \dots, \Psi_n(0, \bar{x}_n, \bar{\mu}_{n1}, \bar{\mu}_{n2})]^{\mathrm{T}}$$
(56)

with  $\bar{\varepsilon}_n := [\varepsilon_1, \ldots, \varepsilon_n]^{\mathrm{T}}$ .

Deringer

*Proof* In the case of  $u \in S_n(\bar{x}_n, \bar{\mu}_{n1}, \bar{\mu}_{n2})$ , we have  $e_{n+1} = 0$  and  $V_{n+1}(e_{n+1}) = 0$ . With the cyclic-small-gain condition (52) satisfied for all i = 1, ..., n, by directly combining the methods in [33,34] and the cyclic-small-gain theorem in [25], as reviewed in Sect. 2, (54) can be proved. This ends the proof.

For specified  $\sigma_i$  for i = 1, ..., n, if we can design the  $\gamma_{e_i}^{\mu_{k1}}$ 's (k = 1, ..., i - 1) and the  $\gamma_{e_i}^{\mu_{k2}}$ 's (k = 1, ..., i - 1) small enough, we can achieve

$$\theta(\bar{\mu}_{n1}, \bar{\mu}_{n2}, \bar{\varepsilon}_n) = \max_{i=1,\dots,n} \left\{ \sigma_i \circ \gamma_{e_i}^{\mu_{i1}}(\mu_{i1}), \sigma_i \circ \gamma_{e_i}^{\mu_{i2}}(\mu_{i2}), \sigma_i(\varepsilon_i) \right\}$$
(57)

for all  $\mu_{i1}, \mu_{i2}, \varepsilon_i > 0$  for i = 1, ..., n.

As pointed out in [15], nested invariant sets with sizes determined by the zooming variables  $\bar{\mu}_{n1}$  and  $\bar{\mu}_{n2}$  are vital for the implementation of dynamic quantization. Lemma 5 summarizes this section by showing the existence of the nested invariant sets for the closed-loop quantized system designed based on Lemmas 1–4.

Define

$$B_{1}(\bar{\mu}_{n1}, \bar{\mu}_{n2}) = \max_{i=1,...,n} \left\{ \begin{array}{l} \sigma_{i} \circ \alpha_{V}(M_{i1}\mu_{i1}), \\ \sigma_{i} \circ \alpha_{V}\left(\frac{1}{1-c_{i1}}\bar{\kappa}_{i}^{-1}(M_{i2}\mu_{i2})\right) \end{array} \right\},$$
(58)

$$B_2(\bar{\mu}_{n1}, \bar{\mu}_{n2}) = \max_{i=1,\dots,n} \left\{ \begin{array}{l} \sigma_i \circ \alpha_V \left( \frac{1}{c_{i1}} \mu_{i1} \right), \\ \sigma_i \circ \alpha_V \left( \frac{1}{1 - c_{i1}} \bar{\kappa}_i^{-1} \left( \frac{1}{c_{i2}} \mu_{i2} \right) \right) \end{array} \right\}.$$
(59)

**Lemma 5** Consider the quantized control system consisting of the plant (11)–(13) and the quantized control law (24)–(25). Under Assumptions 2, 3, 4 and 5, the closedloop quantized system can be transformed into a large-scale system composed of  $e_i$ -subsystems in the form of (41) and (38), and for specified constants  $c_{i1}, c_{i2}$  satisfying (47) for i = 1, ..., n, specified ISS gains  $\gamma_{e_i}^{e_{k'}}$  ( $k \neq k'$ ) satisfying the cyclicsmall-gain condition (52) for all i = 1, ..., n, specified ISS gains  $\gamma_{e_i}^{\mu_{k1}}, \gamma_{e_i}^{\mu_{k2}}$  for i = 1, ..., n, k = 1, ..., i - 1 satisfying (57) and specified arbitrarily small constants  $\varepsilon_i$  for i = 1, ..., n, we can find continuously differentiable, odd, strictly decreasing and radially unbounded functions  $\kappa_i$  for i = 1, ..., n such that (43) holds for i = 1, ..., n. Moreover, if

$$\sigma_{i} \circ \alpha_{V}(M_{i1}\mu_{i1}) = \sigma_{i} \circ \alpha_{V} \left(\frac{1}{1 - c_{i1}}\bar{\kappa}_{i}^{-1}(M_{i2}\mu_{i2})\right)$$
$$= \sigma_{j} \circ \alpha_{V}(M_{j1}\mu_{j1}) = \sigma_{j} \circ \alpha_{V} \left(\frac{1}{1 - c_{j1}}\bar{\kappa}_{j}^{-1}(M_{j2}\mu_{j2})\right)$$
(60)

for all  $i, j = 1, \ldots, n$  and if

$$B_1(\bar{\mu}_{n1}, \bar{\mu}_{n2}) \ge \theta_0 \tag{61}$$

with  $\theta_0 = \max_{i=1,\dots,n} \{\sigma_i(\varepsilon_i)\}$ , then the ISS-Lyapunov function candidate V defined in (53) satisfies

$$B_1(\bar{\mu}_{n1}, \bar{\mu}_{n2}) \ge V(e) \ge \max\{B_2(\bar{\mu}_{n1}, \bar{\mu}_{n2}), \theta_0\}$$
(62)

$$\Rightarrow \max_{\psi \in \Psi(e,x,\bar{\mu}_{n1},\bar{\mu}_{n2})} \nabla V(e)\psi \le -\alpha(V(e)) \tag{63}$$

where  $\Psi$  is defined in (56).

*Proof* Under Assumptions 2 and 5, with Lemma 1, we can transform the closed-loop quantized system into a large-scale system with state e composed of  $e_i$ -subsystems in the form of (41) and (38).

Under Assumptions 2, 3, 4 and 5, by directly using Lemma 2, for any specified constants  $c_{i1}, c_{i2}$  satisfying (47) for i = 1, ..., n, any ISS gains  $\gamma_{e_k}^{e_{k'}}$   $(k \neq k')$  satisfying the cyclic-small-gain condition (52) for all i = 1, ..., n, any specified ISS gains  $\gamma_{e_i}^{\mu_{k1}}, \gamma_{e_i}^{\mu_{k2}}$  for  $i = 1, \dots, n, k = 1, \dots, i-1$  satisfying (57) and specified arbitrarily small constants  $\varepsilon_i$  for i = 1, ..., n, we can find continuously differentiable, odd, strictly decreasing and radially unbounded functions  $\kappa_i$  for  $i = 1, \ldots, n$  such that (43) holds for i = 1, ..., n.

The satisfaction of (47) by appropriately choosing the  $\kappa_i$  for i = 1, ..., n guarantees  $B_1(\bar{\mu}_{n1}, \bar{\mu}_{n2}) > B_2(\bar{\mu}_{n1}, \bar{\mu}_{n2})$  for all positive zooming variables  $\mu_{i1}, \mu_{i2}$ . By using (61), we have  $B_1(\bar{\mu}_{n1}, \bar{\mu}_{n2}) \ge \max\{B_2(\bar{\mu}_{n1}, \bar{\mu}_{n2}), \theta_0\}$ . Recall the definitions of  $V_i(e_i)$  in (40) and V(e) in (53). The equalities in (60) and the left inequality in (62) guarantee (48)–(49). Under Assumption 4, with (47) satisfied, using Lemma 3, we have  $u \in S_n(\bar{x}_n, \bar{\mu}_{n1}, \bar{\mu}_{n2})$ .

Note that (57) is satisfied by appropriately choosing  $\kappa_i$  for  $i = 1, \ldots, n$ . In virtue of (44), (45), (57) and (59),  $\theta(\bar{\mu}_{n1}, \bar{\mu}_{n2}, \bar{\epsilon}_n) = \max\{B_2(\bar{\mu}_{n1}, \bar{\mu}_{n2}), \theta_0\}$ . With the cyclicsmall-gain condition (52) satisfied by appropriately choosing  $\kappa_i$  for  $i = 1, \ldots, n$  and  $u \in S_n(\bar{x}_n, \bar{\mu}_{n1}, \bar{\mu}_{n2})$ , Lemma 4 guarantees the implication in (62) and (63). This ends the proof.

*Remark* 12 Based on Lemma 5, we can appropriately adjust the zooming variables to make conditions (60) and (61) always satisfied such that the sets represented by functions  $B_1$  and  $B_2$  are nested invariant sets. In the next section, we will use the invariant sets to design dynamic quantization logic.

#### 4.5 A guideline for quantized control law design

In this subsection, we provide a guideline to choosing the functions  $\kappa_i$  for the quantized control law (24)–(25) such that the closed-loop quantized system satisfies property (62)–(63). The guideline includes two major steps:

- 1. Choose the ISS parameters of the  $e_i$ -subsystems.

  - (a) Choose constants c<sub>i1</sub>, c<sub>i2</sub> to satisfy (47) for i = 1, ..., n;
    (b) Choose ISS gains γ<sub>ei</sub><sup>ej</sup> ∈ ℋ<sub>∞</sub> (j ≠ i) and the corresponding functions  $\hat{\gamma}_{e_i}^{e_j} > \gamma_{e_i}^{e_j}$  to satisfy the cyclic-small-gain condition (52) for all i = 1, ..., n, and calculate the  $\sigma_i$  for i = 1, ..., n in (53);

- (c) Choose ISS gains  $\gamma_{e_i}^{\mu_{k1}}$ ,  $\gamma_{e_i}^{\mu_{k2}}$  for  $i = 1, \ldots, n, k = 1, \ldots, i 1$  such that (57) holds for all  $\mu_{i1}, \mu_{i2}, \varepsilon_i > 0$  for  $i = 1, \ldots, n$ ;
- (d) Choose specified  $\varepsilon_i$ ,  $\iota_i > 0$  for i = 1, ..., n.
- 2. Choose the  $\kappa_i$  for i = 1, ..., n based on Lemma 2 with the ISS parameters chosen in Step 1.

*Remark 13* In Step 1, it is only required that Step (c) is after Step (b), because condition (60) in Step (c) depends on the  $\sigma_i$  calculated in Step (b). Under Assumptions 2–5, if the ISS parameters and the  $\kappa_i$  are chosen according to the guideline and if conditions (60) and (61) are satisfied, then from Lemma 5, the nested invariant sets exist.

## **5** Dynamic quantization

Because of the saturation property of the quantizers, the quantized control law designed in Sect. 4 can only guarantee local stabilization; see (62)–(63). In this section, based on the invariant sets given in Lemma 5, we design a dynamic quantization logic in the form of (30), composed of zooming-in stage and zooming-out stage, to dynamically adjust the zooming variables  $\mu_{ij}$  (i = 1, ..., n, j = 1, 2) such that the closed-loop quantized system is globally stabilized. Recall that in dynamic quantization, the zooming variables  $\mu_{ij}(t)$  are piecewise constant signals, and are adjusted on a discrete time sequence { $t_0, t_1, t_2, ...$ } where  $t_{k+1} - t_k = t_d$  with constant  $t_d > 0$ .

To satisfy condition (60) in Lemma 5, we design dynamic quantization such that for all  $t \in \mathbb{R}_+$ 

$$\sigma_i \circ \alpha_V(M_{i1}\mu_{i1}(t)) = \sigma_i \circ \alpha_V \left(\frac{1}{1 - c_{i1}}\bar{\kappa}_i^{-1}(M_{i2}\mu_{i2}(t))\right) := \Theta(t)$$
(64)

for i = 1, ..., n. Equivalently, we have

$$\mu_{i1}(t) = \frac{1}{M_{i1}} \alpha_V^{-1} \circ \sigma_i^{-1}(\Theta(t)) := \Upsilon_{i1}(\Theta(t)),$$
(65)

$$\mu_{i2}(t) = \frac{1}{M_{i2}} \bar{\kappa}_i \left( (1 - c_{i1}) \alpha_V^{-1} \circ \sigma_i^{-1}(\Theta(t)) \right) := \Upsilon_{i2}(\Theta(t))$$
(66)

for i = 1, ..., n. Note that  $\Upsilon_{i1}$  and  $\Upsilon_{i2}$  are invertible for i = 1, ..., n. Thus, the dynamic quantization logic (30) can be designed by choosing an appropriate update law for  $\Theta$ , which can largely reduce the design complexity for all the zooming variables  $\mu_{ij}$  (i = 1, ..., n, j = 1, 2). The update law for  $\Theta$  is expected to be in the following form:

$$\Theta(t_{k+1}) = Q(\Theta(t_k)), \quad k \in \mathbb{Z}_+$$
(67)

where  $t_{k+1} - t_k = t_d$  with constant  $t_d > 0$ . In the zooming-out stage,  $Q = Q^{\text{out}}$ ; in the zooming-in stage,  $Q = Q^{\text{in}}$ . With  $Q^{\text{out}}$  and  $Q^{\text{in}}$  designed, we can design the

dynamic quantization logic (30) for  $\mu_{ij}$  by choosing

$$Q_{ij}^{\text{out}} = \Upsilon_{ij} \circ Q^{\text{out}} \circ \Upsilon_{ij}^{-1}, \quad Q_{ij}^{\text{in}} = \Upsilon_{ij} \circ Q^{\text{in}} \circ \Upsilon_{ij}^{-1}.$$
(68)

Using the definition of  $B_1$  in (58), we also have

$$\Theta(t) = B_1(\bar{\mu}_{n1}(t), \bar{\mu}_{n2}(t)).$$
(69)

Before designing dynamic quantization, the relation between zooming variables  $\bar{\mu}_{n1}$ ,  $\bar{\mu}_{n2}$  and control error *e* should be clarified. For i = 1, ..., n, using the definitions of  $S_i$  in (36), the strictly decreasing property of  $\kappa_i$  implies

$$\max S_i(\bar{x}_i, \bar{\mu}_{i1}, \bar{\mu}_{i2}) = \kappa_i(x_i - \max S_{i-1}(\bar{x}_{i-1}, \bar{\mu}_{(i-1)1}, \bar{\mu}_{(i-1)2}) - \max\{c_{i1}|e_i|, \mu_{i1}\}) + \mu_{i2},$$
(70)

and

$$\min S_i(\bar{x}_i, \bar{\mu}_{i1}, \bar{\mu}_{i2}) = \kappa_i (x_i - \min S_{i-1}(\bar{x}_{i-1}, \bar{\mu}_{(i-1)1}, \bar{\mu}_{(i-1)2}) + \max\{c_{i1}|e_i|, \mu_{i1}\}) - \mu_{i2}.$$
(71)

Recall  $e = [e_0^T, e_1, \dots, e_n]^T$ . Given the definitions of  $e_i$  for  $i = 2, \dots, n$  in (37), we can denote

$$e = e(X, \bar{\mu}_{n1}, \bar{\mu}_{n2})$$
 (72)

with  $X = [z^{T}, x^{T}]^{T} \in \mathbb{R}^{n+n_{z}}$ . It can be observed that *e* is a continuous function of  $X, \bar{\mu}_{n1}, \bar{\mu}_{n2}$ . Clearly, the piecewise constant adjustment of  $\bar{\mu}_{n1}, \bar{\mu}_{n2}$  causes jumps of *e* on the time-line. This makes a difference from the previously known results.

## 5.1 Zooming-out stage

The design of the zooming-out stage is motivated by [21]. The purpose of the zoomingout stage in this subsection is to increase the zooming variables  $\mu_{ij}$  such that at some finite time  $t_{k^*}$ , the state of the closed-loop quantized system is in the larger invariant set corresponding to  $B_1$  in (62). In this stage, the components  $\kappa_i$  for i = 1, ..., n of the controller are set to be zero. Thus, u = 0.

The small-time norm-observability assumed in Assumption 1 guarantees that for  $t_d > 0$ , there exists a  $\varphi \in \mathscr{K}_{\infty}$  such that

$$|X(t_k + t_d)| \le \varphi(\|x\|_{[t_k, t_k + t_d]})$$
(73)

for any  $k \in \mathbb{Z}_+$ . Considering the definitions of *V* and *e* in (53) and (72), for  $t_d > 0$ , property (73) can be represented with the Lyapunov function *V* as

$$|V(e(X(t_k + t_d), 0, 0))| \le \bar{\varphi}(||x||_{[t_k, t_k + t_d]})$$
(74)

🖉 Springer

for any  $k \in \mathbb{Z}_+$ , where  $\bar{\varphi} \in \mathscr{K}_{\infty}$ .

With the forward completeness property assumed in Assumption 1, we design a zooming-out logic  $Q^{\text{out}}$ :  $\mathbb{R}_+ \to \mathbb{R}_+$  to increase  $\Theta$  fast enough to dominate the growth rate of  $\overline{\varphi}(|x|)$  such that at some finite time  $t_{k^*} > 0$  with  $k^* \in \mathbb{Z}_+$ , it holds:

$$M_{i1}\mu_{i1}(t_{k^*}) \ge |x_i(t_{k^*})|, \quad i = 1, \dots, n,$$
(75)

$$\Theta(t_{k^*}) \ge \bar{\varphi}(\|x\|_{[t_{k^*} - t_d, t_{k^*}]}).$$
(76)

*Remark 14* Due to the saturation of the quantizer, if the input signal of a quantizer is outside the range of the quantizer, then we cannot estimate the bound of the signal. In the zooming-out stage, the  $\kappa_i$ 's are set to be zero, and the input of the quantizer  $q_{i1}$  is  $x_i$ ; see control law (24)–(25). Inequality (75) means that at some finite time  $t_{k^*}$ ,  $x_i$  is in the quantization range of  $q_{i1}$ . Then, we can estimate the bound of  $|x_i(t_{k^*})|$ .

Using (74) and (76), we have

$$\Theta(t_{k^*}) \ge \max\{V(e(X(t_{k^*}), 0, 0)), \theta_0\}.$$
(77)

From the definitions of max  $S_i(\bar{x}_i, \bar{\mu}_{i1}, \bar{\mu}_{i2})$ , min  $S_i(\bar{x}_i, \bar{\mu}_{i1}, \bar{\mu}_{i2})$  and  $e_{i+1}$ , one observes that increase of  $\bar{\mu}_{n1}, \bar{\mu}_{n2}$  leads to increase of max  $S_i(\bar{x}_i, \bar{\mu}_{i1}, \bar{\mu}_{i2})$ , decrease of min  $S_i(\bar{x}_i, \bar{\mu}_{i1}, \bar{\mu}_{i2})$  and thus decrease or hold of  $|e_{i+1}|$  for i = 1, ..., n-1. Thus, with the zooming-out logic  $Q^{\text{out}}$ , we achieve that, at time  $t_{k^*} > 0$  with  $k^* \in \mathbb{Z}_+$ , it holds that

$$\Theta(t_{k^*}) \ge \max\{V(e(X(t_{k^*}), \bar{\mu}_{n1}(t_{k^*}), \bar{\mu}_{n2}(t_{k^*}))), \theta_0\}.$$
(78)

With  $Q^{\text{out}}$  designed, we can design the zooming-out logic  $Q_{ij}^{\text{out}}$  for i = 1, ..., n, j = 1, 2 according to (68).

*Remark 15* As mentioned in Remark 2, if a bound of the initial state X(0) is known, we can directly set  $\Theta(0)$  to satisfy (77) with  $t_{k^*} = 0$ . In this case, the zooming-out stage is not necessary and Assumption 1 is not required.

#### 5.2 Zooming-in stage

The zooming-out stage achieves (78) at time  $t_{k^*}$  with  $k^* \in \mathbb{Z}_+$ . Suppose that at some  $t_k > 0$  with  $k \ge k^*$ ,

$$\Theta(t_k) \ge \max\{V(e(X(t_k), \bar{\mu}_{n1}(t_k), \bar{\mu}_{n2}(t_k))), \theta_0\}.$$
(79)

In this subsection, first we design a  $Q^{in} : \mathbb{R}_+ \to \mathbb{R}_+$  for the zooming-in stage such that

$$\Theta(t_{k+1}) = Q^{\mathrm{in}}(\Theta(t_k)) \ge \max\{V(e(X(t_{k+1}), \bar{\mu}_{n1}(t_{k+1}), \bar{\mu}_{n2}(t_{k+1}))), \theta_0\}.$$
 (80)

This objective will be achieved by Lemmas 6 and 7. Then we show the convergence property of the update law (67) for  $\Theta$  in the zooming-in stage by Lemma 8.

*Remark 16* If (80) is achieved based on (79), we can recursively guarantee that the state *e* of the closed-loop quantized system is always in the larger invariant set represented by  $B_1$  in spite of the update of  $\Theta$ ; see (62) and (69).

The following lemma describes the decreasing property of V during the time interval  $[t_k, t_{k+1})$ , based on which we will design the zooming-in update law  $Q^{\text{in}}$  for  $\Theta$ .

**Lemma 6** Consider the closed-loop quantized system with V satisfying property (62)–(63). If (79) holds at time  $t_k$  with  $k \in \mathbb{Z}_+$ , then there exists a continuous and positive definite function  $\bar{\rho}$  such that

$$(\mathrm{Id} - \bar{\rho}) \in \mathscr{K}_{\infty},\tag{81}$$

$$V(e(X(t_{k+1}), \bar{\mu}_{n1}(t_k), \bar{\mu}_{n2}(t_k))) \le \max\{(\mathrm{Id} - \bar{\rho})(\Theta(t_k)), \theta_0\}.$$
(82)

The proof of Lemma 6 is in Appendix D.

From the definition in (72), the piece-wise constant update of the zooming variables  $\bar{\mu}_{n1}$ ,  $\bar{\mu}_{n2}$  causes jumps of *e* and thus jumps of *V*. Based on (82), we design the zooming-in logic  $Q^{\text{in}}$  to achieve (80) in spite of the jumps.

To clearly represent the relation between Lyapunov function V and X,  $\Theta$ , define

$$W(\xi, s) = V(e(\xi, \Upsilon_{n1}(s), \Upsilon_{n2}(s)))$$
(83)

for  $\xi \in \mathbb{R}^{n+n_z}$  and  $s \in \mathbb{R}_+$ , where

$$\overline{\Upsilon}_{n1}(s) = [\Upsilon_{11}(s), \dots, \Upsilon_{n1}(s)]^{\mathrm{T}}, \tag{84}$$

$$\overline{\Upsilon}_{n1}(s) = [\Upsilon_{12}(s), \dots, \Upsilon_{n2}(s)]^{\mathrm{T}}.$$
(85)

Then,  $W(\xi, s)$  is a continuous function of  $(\xi, s)$ .

Consider  $(\xi, s)$  satisfying

$$0 \le s \le \Theta(t_{k^*}) \tag{86}$$

$$W(\xi, s) \le \Theta(t_{k^*}). \tag{87}$$

From the definitions of *V* and *W* in (53) and (83), we can find a compact set  $\Omega^o \subset \mathbb{R}^{n+n_z} \times \mathbb{R}_+$  such that all the  $(\xi, s)$  satisfying (86)–(87) belong to  $\Omega^o$ . By using the property of continuous functions, we can find a continuous and positive definite function  $\rho^o < \text{Id}$  such that for all  $(\xi, s) \in \Omega^o$  and all  $h \ge 0$ , it holds that

$$|W(\xi, s - \rho^{o}(h)) - W(\xi, s)| \le h.$$
(88)

We propose the following update law for  $\Theta$  in the zooming-in stage:

$$Q^{\text{in}}(\Theta) = \Theta - \rho^o \left(\frac{\Theta - \max\{\Xi(\Theta), \theta_0\}}{2}\right).$$
(89)

where  $\Xi = (Id - \bar{\rho})$ . In the following procedure, we employ Lemma 7 to guarantee objective (80) and employ Lemma 8 to show the convergence property of  $\Theta$  with update law defined in (89).

Lemma 7 shows that property (80) can be achieved with the zooming-in update law (89) for  $\Theta$ , given (78) and (79) satisfied.

**Lemma 7** Consider the closed-loop quantized system with V satisfying property (62)– (63). Suppose that condition (78) holds at some finite time  $t_{k*}$  and condition (79) holds at some time  $t_k$  with  $k \ge k^*$ . Then, property (80) is satisfied at time  $t_{k+1}$  with the update law  $\Theta(t_{k+1}) = Q^{in}(\Theta(t_k))$  with  $Q^{in}$  defined in (89).

*Proof* With  $0 < \rho^o < \text{Id}$ , it can be guaranteed that

$$\Theta(t_{k+1}) \le \Theta(t_k) \tag{90}$$

and

$$\Theta(t_{k+1}) = \Theta(t_k) - \rho^o \left( \frac{\Theta(t_k) - \max\{\Xi(\Theta(t_k)), \theta_0\}}{2} \right)$$
  
$$\geq \frac{\Theta(t_k) + \max\{\Xi(\Theta(t_k)), \theta_0\}}{2}$$
(91)

for  $k \ge k^*$ . Thus,  $0 < \Theta(t_k) \le \Theta(t_{k^*})$  for  $k \ge k^*$ .

From Lemma 6, (82) holds. Using (78), (82) and (83), we have

$$W(X(t_{k+1}), \Theta(t_k)) \le \max\{\Xi(\Theta(t_k)), \theta_0\} \le \Theta(t_{k^*})$$
(92)

for  $k \ge k^*$ . Hence,  $(X(t_{k+1}), \Theta(t_k)) \in \Omega^o$  for  $k \ge k^*$ . Given  $(X(t_{k+1}), \Theta(t_k)) \in \Omega^o$ , from (89) and (92), we obtain

$$W(X(t_{k+1}), \Theta(t_{k+1})) \leq W(X(t_{k+1}), \Theta(t_{k+1})) - W(X(t_{k+1}), \Theta(t_{k}))| \leq W(X(t_{k+1}), \Theta(t_{k})) + |W(X(t_{k+1}), Q^{in}(\Theta(t_{k})) - W(X(t_{k+1}), \Theta(t_{k}))| \leq \max\{\Xi(\Theta(t_{k})), \theta_{0}\} + \frac{\Theta(t_{k}) - \max\{\Xi(\Theta(t_{k})), \theta_{0}\}}{2} = \frac{\Theta(t_{k}) + \max\{\Xi(\Theta(t_{k})), \theta_{0}\}}{2}.$$
(93)

From (79), we have  $\theta_0 \leq \Theta(t_k)$ , which implies

$$\theta_0 \le \frac{\Theta(t_k) + \max\{\Xi(\Theta(t_k)), \theta_0\}}{2}.$$
(94)

Properties (91), (93) and (94) together with the definition of W in (83) guarantee (80). This ends the proof of Lemma 7.

Lemma 8 shows the convergence property of the update law (67) for  $\Theta$  with  $Q = Q^{\text{in}}$  defined in (89). The proof of Lemma 8 can be found in [44, Section IV.C].

**Lemma 8** Suppose that at some  $t_{k^*} > 0$  with  $k^* \in \mathbb{Z}_+$ ,  $\Theta(t_{k^*}) \ge \theta_0$ . Then with  $Q^{\text{in}}$  defined in (89), update law  $\Theta(t_{k+1}) = Q^{\text{in}}(\Theta(t_k))$  achieves

$$\lim_{k \to \infty} \Theta(t_k) = \theta_0.$$
<sup>(95)</sup>

*Remark 17* The zooming-in update law for  $\Theta$  can be designed by finding the  $\bar{\rho}$  with Lemma 6 and  $\rho^o$  by using the continuity of W. Lemmas 7 and 8 are used to prove the effectiveness of the the zooming-in update law  $Q^{\text{in}}$  defined in (89). With  $Q^{\text{in}}$  designed, we can design the zooming-in logic  $Q_{ii}^{\text{in}}$  for i = 1, ..., n, j = 1, 2 according to (68).

#### 5.3 Main result

The main result of quantized control is summarized in Theorem 2.

**Theorem 2** For the system (11)–(13), under Assumptions 1–4, by choosing constants  $c_{i1}, c_{i2}$  satisfying (47) for i = 1, ..., n, ISS gains  $\gamma_{e_k}^{e_{k'}}$  ( $k \neq k'$ ) satisfying the cyclic-small-gain condition (52) for all i = 1, ..., n, ISS gains  $\gamma_{e_i}^{\mu_{k1}}, \gamma_{e_i}^{\mu_{k2}}$  for i = 1, ..., n, k = 1, ..., i - 1 satisfying (57) and constants  $\varepsilon_i > 0$  for i = 1, ..., n, we can design the functions  $\kappa_i$  for i = 1, ..., n in (24)–(25) and the dynamic quantization logic  $Q_{ij}$  for i = 1, ..., n, j = 1, 2 in (30) such that the closed-loop solutions z and x are globally bounded. Moreover, by choosing the constants  $\varepsilon_i > 0$  for i = 1, ..., n arbitrarily small, the output  $x_1(t)$  can be steered to within an arbitrarily small neighborhood of origin.

*Proof* With Assumption 1 satisfied, at some time  $t_{k^*} > 0$  with  $k^* \in \mathbb{Z}_+$ , (78) can be achieved by the zooming-out logic  $Q_{ij}^{\text{out}}$  designed in Sect. 5.1.

With Assumptions 2–4 satisfied, using Lemma 5, by appropriately designing the functions  $\kappa_i$  for i = 1, ..., n such that the ISS parameters satisfy the conditions (47), (52) and (57), the closed-loop quantized system has the nested invariant sets defined in (63).

Using Lemmas 6 and 7, (80) can be guaranteed with the zooming-in logic designed in Sect. 5.2, and it holds that  $V(\underline{e}(X(t_k), \bar{\mu}_{n1}(t_k), \bar{\mu}_{n2}(t_k))) \leq \Theta(t_k)$  for  $k \geq k^*$ . Moreover, Lemma 8 implies that  $\overline{\lim}_{k\to\infty} V(\underline{e}(X(t_k), \bar{\mu}_{n1}(t_k), \bar{\mu}_{n2}(t_k))) \leq \theta_0$ . Recall the definition of V in (53). The closed-loop signal  $x_1$  is driven to within the region  $|x_1| \leq \alpha_V^{-1}(\theta_0)$ . Recall the definition of  $\theta_0$  in Lemma 5. By designing  $\varepsilon_i$  (i = 1, ..., n) arbitrarily small, the state  $x_1$  can be steered to within an arbitrarily small neighborhood of the origin. This ends the proof.

*Remark 18* The main result of the paper is still new when there is no inverse dynamics (i.e., the *z*-subsystem) in system (11)–(13). In this case, the assumption on small-time norm-observability in Assumption 1 is not needed.

#### 6 A remark on robust measurement feedback control

Consider system (11)–(13) under additive state measurement disturbances without quantization. For i = 1, ..., n, denote  $x_i^m$  as the measurement of  $x_i$  and  $w_i$  as the corresponding measurement disturbance, that is,  $x_i^m = x_i + w_i$ . We show that the above designed procedure can be adopted to solve the robust measurement feedback control problem.

**Assumption 6** For each i = 1, ..., n, there exists a constant  $\mu_{i1} \ge 0$  such that

$$\|w_i\| \le \mu_{i1}.\tag{96}$$

Under Assumptions 2, 3 and 6, robust control with disturbed measurements can be addressed in a similar way to the quantized control case. In the constructive control design, let  $\mu_{i2} = 0$  for i = 1, ..., n. The measurement feedback control law is designed recursively as

и

$$v_i = \kappa_i (x_i^m - v_{i-1}) = \kappa_i (x_i - v_{i-1} + w_i), \quad i = 1, \dots, n$$
(97)

$$=v_n$$
 (98)

with  $v_0 = 0$ . It can be directly checked that

$$v_1 \in S_1(\bar{x}_1, \bar{\mu}_{11}, \bar{\mu}_{12}) \Rightarrow \dots \Rightarrow v_i \in S_i(\bar{x}_i, \bar{\mu}_{i1}, \bar{\mu}_{i2}) \Rightarrow \dots \Rightarrow v_n \in S_n(\bar{x}_n, \bar{\mu}_{n1}, \bar{\mu}_{n2}).$$
(99)

An ISS-Lyapunov function V(e) constructed as in (53) with  $\theta$  defined as in (57) satisfies

$$V(e) \ge \theta(\bar{\mu}_{n1}, \bar{\mu}_{n2}, \bar{\varepsilon}_n) \Rightarrow \max_{\psi \in \Psi(e, x, \bar{\mu}_{n1}, \bar{\mu}_{n2})} \nabla V(e)\psi \le -\alpha(V(e))$$
(100)

wherever  $\nabla V$  exists, where  $\alpha$  is continuous and positive definite, and  $\Psi$  is defined in (56). Note that  $\sigma_1 = \text{Id}$ . We can then achieve  $\theta(\bar{\mu}_{n1}, \bar{\mu}_{n2}, \bar{\epsilon}_n) = \gamma_{e_1}^{\mu_{11}}(\mu_{11}) = \alpha_V(\mu_{11}/c_{11})$  by designing  $\sigma_i$  (i = 2, ..., n) and  $\epsilon_i$  (i = 1, ..., n) small enough in the recursive design procedure. This together with (100) means that V(e) ultimately converges to a region  $V(e) \leq \alpha_V(\mu_{11}/c_{11})$ . It can be derived that  $x_1$  ultimately converges to the region  $|x_1| \leq \mu_{11}/c_{11}$ . With  $c_{11}$  chosen close to one in the constructive design procedure, the state  $x_1$  ultimately converges to the region close to  $|x_1| \leq \mu_{11}$ .

Without a detailed proof, we present a theorem on the robust measurement feedback control for system (11)–(13).

**Theorem 3** Consider system (11)–(13). Under Assumptions 2, 3 and 6, the closedloop signals are bounded, and in particular, the state  $x_1$  can be steered to within a region arbitrarily close to  $|x_1| \le \mu_{11}$  with the measurement feedback control law in (97)–(98).

### 7 Summary and future work

This paper has developed a new tool for quantized nonlinear control design. The essential strategy is to introduce a change of coordinates so that the closed-loop quantized system is a large-scale system of ISS subsystems described by differential inclusions. The recently developed cyclic-small-gain theorem is employed to guarantee the stability property of the closed-loop quantized system and to construct an ISS-Lyapunov function to evaluate the influence of quantization errors. One significance of this result is that an *n*-dimensional strict-feedback nonlinear system with measurement and actuator quantization can be globally stabilized by a quantized controller with 2n three-level dynamic quantizers.

In spite of the obtained results, several related problems should be addressed in the future research:

- Quantized control is closely related to other network control problems such as sampled-data control, control with time-delays. How to deal with more complex network behaviors in a systematic way, in particular those hybrid/switching systems satisfying only a weak semigroup property (see [45]), should be studied in greater details.
- The cyclic-small-gain theorem was originally developed for large-scale systems.
   It is thus very natural to ask whether decentralized quantized controllers can be developed for a class of large-scale nonlinear systems.
- Controllers are expected to possess adaptive capabilities to cope with "large" system uncertainties. A further extension of the presented methodology to quantized adaptive control is of practical interest for engineering applications.

#### **Appendix: Proofs of technical Lemmas**

#### A Proof of Lemma 1

We simply use  $S_k$  instead of  $S_k(\bar{x}_k, \bar{\mu}_{k1}, \bar{\mu}_{k2})$  for  $k = 1, \dots, i - 1$ .

We only consider the case of  $e_i > 0$ .

Consider the recursive definition of  $S_k$ 's in (36). For k = 1, ..., i - 1, the strictly decreasing property of the  $\kappa_k$ 's implies

 $\max S_k = \kappa_k (x_k - \max S_{k-1} - \max\{c_{k1}|e_k|, \mu_{k1}\}) + \mu_{k2}, \tag{101}$ 

$$\min S_k = \kappa_k (x_k - \min S_{k-1} + \max\{c_{k1}|e_k|, \mu_{k1}\}) - \mu_{k2}.$$
(102)

From the iteration type definition of  $e_k$ 's for  $k = 1, ..., i - 1, e_{i-1}$  is continuous and differentiable almost everywhere with respect to  $\bar{x}_{i-1}, \bar{\mu}_{(i-2)1}, \bar{\mu}_{(i-2)2}$ .

Since  $\kappa_k$ 's are continuously differentiable for k = 1, ..., i - 1, using (101), we can see max  $S_{i-1}$  is continuously differentiable almost everywhere with respect to  $\bar{x}_{i-1}, \bar{\mu}_{(i-1)1}, \bar{\mu}_{(i-1)2}$ .

Considering the definition of  $e_i$  in (37), when  $e_i > 0$ , we can represent the  $e_i$ -subsystem with a differential equation

$$\dot{e}_i = x_{i+1} + \Delta_i(\bar{x}_i, z) - \nabla \max S_{i-1}[\bar{x}_{i-1}, 0_{(i-1)}, 0_{(i-1)}]^{\mathrm{T}}$$
(103)

with  $0_{(i-1)}$  being the vector composed of i - 1 zero elements, wherever max  $S_{i-1}$  is continuously differentiable, or equivalently,  $\nabla \max S_{i-1}$  exists. Because max  $S_{i-1}$  is continuously differentiable almost everywhere,  $\nabla \max S_{i-1}$  is discontinuous and thus the  $e_i$ -subsystem is a discontinuous system. We represent the  $e_i$ -subsystem with a differential inclusion by embedding the discontinuous  $\nabla \max S_{i-1}$  into a set-valued map

$$\partial \max S_{i-1} = \bigcap_{\varepsilon > 0} \bigcap_{\tau(\tilde{\mathcal{M}}) = 0} \overline{\operatorname{co}} \left\{ \nabla \max S_{i-1}(\mathscr{B}_{\varepsilon}(\zeta_{i-1}) \setminus \tilde{\mathcal{M}}) \right\}$$
(104)

where  $\mathscr{B}_{\varepsilon}(\zeta_{i-1})$  is an open ball of radius  $\varepsilon$  around  $\zeta_{i-1} := [\bar{x}_{i-1}^{\mathrm{T}}, \bar{\mu}_{(i-1)1}^{\mathrm{T}}, \bar{\mu}_{(i-1)2}^{\mathrm{T}}]^{\mathrm{T}}$ , and  $\tilde{\mathscr{M}}$  represents all sets of zero measure (i.e.,  $\tau(\tilde{\mathscr{M}}) = 0$ ). Then,  $\partial \max S_{i-1}$  is convex, compact and upper semi-continuous (see [46, Chapter 1] for a tutorial and [33,34] for recent results on such properties of discontinuous systems).

Then, in the case of  $e_i > 0$ , the  $e_i$ -subsystem can be represented with a differential inclusion as

$$\dot{e}_{i} \in \{x_{i+1} + \Delta_{i}(\bar{x}_{i}, z) - \varphi_{i} : \varphi_{i} \in \partial \max S_{i-1}[\bar{x}_{i-1}^{1}, 0_{(i-1)}, 0_{(i-1)}]^{1}\} \\ := \{x_{i+1} + \varphi_{i} : \varphi_{i} \in \Phi_{i}(\bar{x}_{i}, \bar{\mu}_{(i-1)1}, \bar{\mu}_{(i-1)2}, z)\}$$
(105)

where

$$\Phi_{i}(\bar{x}_{i}, \bar{\mu}_{(i-1)1}, \bar{\mu}_{(i-1)2}, z) = \{\Delta_{i}(\bar{x}_{i}, z) - \varphi_{i} : \varphi_{i} \in \partial \max S_{i-1}[\dot{\bar{x}}_{i-1}^{\mathrm{T}}, 0_{(i-1)}, 0_{(i-1)}]^{\mathrm{T}}\}.$$
 (106)

Because  $\Delta_i$  and  $\dot{\bar{x}}_i$  are locally Lipschitz and  $\partial \max S_{i-1}$  is convex, compact and upper semi-continuous,  $\Phi_i$  is convex, compact and upper semi-continuous. Considering the definition of  $\partial \max S_{i-1}$ , one can find a continuous function  $\bar{s}_{i-1}$  such that for all  $\bar{x}_{i-1}$ ,  $\bar{\mu}_{(i-1)1}$ ,  $\bar{\mu}_{(i-1)2}$ , any  $s_{i-1} \in \partial \max S_{i-1}$  satisfies  $|s_{i-1}| \leq \bar{s}_{i-1}(\bar{x}_{i-1}, \bar{\mu}_{(i-1)1}, \bar{\mu}_{(i-1)2}, z)$ . Thus, for all  $\bar{x}_{i-1}$ ,  $\bar{\mu}_{(i-1)1}$ ,  $\bar{\mu}_{(i-1)2}$ , z, any  $\phi_i \in \Phi_i(\bar{x}_i, \bar{\mu}_{(i-1)1}, \bar{\mu}_{(i-1)2}, z)$  satisfies

$$|\phi_i| \le |\Delta_i(\bar{x}_i, z)| + \bar{s}_{i-1}(\bar{x}_{i-1}, \bar{\mu}_{(i-1)1}, \bar{\mu}_{(i-1)2})|\bar{x}_{i-1}|.$$
(107)

From (12)–(13) and Assumption 2,  $\Delta_i(\bar{x}_i, z)$  is bounded by a  $\mathscr{K}_{\infty}$  function of  $(\bar{x}_i, z)$ and  $\dot{\bar{x}}_{i-1}$  is bounded by a  $\mathscr{K}_{\infty}$  function of  $(\bar{x}_i, z)$ . Thus, there exists a  $\lambda_{\Phi_i}^0 \in \mathscr{K}_{\infty}$  such that for any  $\phi_i \in \Phi_i(\bar{x}_i, \bar{\mu}_{(i-1)1}, \bar{\mu}_{(i-1)2}, z)$ , it holds that

$$|\phi_i| \le \lambda_{\Phi_i}^0(|(\bar{x}_i, \bar{\mu}_{(i-1)1}, \bar{\mu}_{(i-1)2}, z)|).$$
(108)

Deringer

For the purpose of (39), for each k = 1, ..., i - 1, we find a  $\lambda_{x_{k+1}} \in \mathscr{K}_{\infty}$  such that  $|x_{k+1}| \leq \lambda_{x_{k+1}}(|(\bar{e}_{k+1}, \bar{\mu}_{k1}, \bar{\mu}_{k2})|)$ . For k = 1, ..., i - 1, from the definitions of  $x_{k+1}$  in (37), we have min  $S_k \leq x_{k+1} - e_{k+1} \leq \max S_k$  and thus

$$|x_{k+1}| \le \max\{|\max S_k|, |\min S_k|\} + |e_{k+1}|.$$
(109)

For each k = 1, ..., i - 1, define  $\kappa_k^o(s) = \kappa_k(|s|)$  for  $s \in \mathbb{R}_+$ . Because  $\kappa_k$  is odd, strictly decreasing and radially unbounded,  $\kappa_k^o \in \mathscr{K}_\infty$ . From (101), we have

$$|\max S_{k}| \leq \kappa_{k}^{o}(|x_{k} - \max S_{k-1} - \max\{c_{k1}|e_{k}|, \mu_{k1}\}|) + \mu_{k2}$$
  
$$\leq \kappa_{k}^{o}(|x_{k} - \max S_{k-1}| + \max\{c_{k1}|e_{k}|, \mu_{k1}\}) + \mu_{k2}$$
  
$$\leq \kappa_{k}^{o}(|\max S_{k-1}| + |\min S_{k-1}| + |e_{k}| + \max\{c_{k1}|e_{k}|, \mu_{k1}\}) + \mu_{k2}.$$
(110)

In (110), we used the fact that min  $S_{k-1} \le x_k - e_k \le \max S_{k-1}$  and thus min  $S_{k-1} - \max S_{k-1} + e_k \le x_k - \max S_{k-1} \le e_k$ , to arrive at  $|x_k - \max S_{k-1}| \le |\max S_{k-1}| + |\min S_{k-1}| + |e_k|$ . Similarly, we obtain:

$$|\min S_k| \le \kappa_k^o(|\max S_{k-1}| + |\min S_{k-1}| + |e_k| + \max\{c_{k1}|e_k|, \mu_{k1}\}) + \mu_{k2}.$$
(111)

For each  $x_{k+1}$  (k = 1, ..., i - 1), using (109), (110) and (111), one can find a  $\lambda_{x_{k+1}} \in \mathscr{H}_{\infty}$  such that  $|x_{k+1}| \leq \lambda_{x_{k+1}}(|(\bar{e}_{k+1}, \bar{\mu}_{k1}, \bar{\mu}_{k2})|)$ . This together with (108) guarantees that there exists a  $\lambda_{\Phi_i} \in \mathscr{H}_{\infty}$  such that for all  $(\bar{x}_i, \bar{\mu}_{(i-1)1}, \bar{\mu}_{(i-1)2}, z)$ , any  $\phi_i \in \Phi_i(\bar{x}_i, \bar{\mu}_{(i-1)1}, \bar{\mu}_{(i-1)2}, z)$  satisfies

$$|\phi_i| \le \lambda_{\Phi_i}(|(\bar{e}_i, z, \bar{\mu}_{(i-1)1}, \bar{\mu}_{(i-1)2})|)$$
(112)

where  $\bar{e}_i := [e_1, \ldots, e_i]^{\mathrm{T}}$ . Define

$$\Phi_i^*(e_{i+1}, \bar{x}_i, \bar{\mu}_{(i-1)1}, \bar{\mu}_{(i-1)2}, z) = \{\phi_i + e_{i+1} : \phi_i \in \Phi_i(\bar{x}_i, \bar{\mu}_{(i-1)1}, \bar{\mu}_{(i-1)2}, z)\}.$$
(113)

From (37),  $x_{i+1} - e_{i+1} \in S_i(\bar{x}_i, \bar{\mu}_{i1}, \bar{\mu}_{i2})$ . Then, equation (105) can be rewritten as (38).

From (112), we can find a  $\lambda_{\Phi_i^*} \in \mathscr{K}_{\infty}$  such that for all  $(e_{i+1}, \bar{x}_i, \bar{\mu}_{(i-1)1}, \bar{\mu}_{(i-1)2}, z)$ , any  $\phi_i^* \in \Phi_i^*(e_{i+1}, \bar{x}_i, \bar{\mu}_{(i-1)1}, \bar{\mu}_{(i-1)2}, z)$  satisfies (39).

By also considering the cases of  $e_i = 0$  and  $e_i < 0$ , Lemma 1 can be proved. This ends the proof.

## B Proof of Lemma 2

Note that  $e_0 = z$ . With (39) satisfied, one can find  $\lambda_{\Phi_i^*}^{e_k} \in \mathscr{H}_\infty$  for  $k = 0, \ldots, i+1$  and  $\lambda_{\Phi_i^*}^{\mu_{k1}}, \lambda_{\Phi_i^*}^{\mu_k} \in \mathscr{H}_\infty$  for  $k = 1, \ldots, i-1$  such that for any  $\phi_i^* \in \Phi_i^*(e_{i+1}, \bar{x}_i, \bar{\mu}_{(i-1)1}, \bar{\mu}_{(i-1)2}, z)$ , it holds that

$$|\phi_i^*| \le \sum_{k=1}^{i+1} \lambda_{\Phi_i^*}^{e_k}(|e_k|) + \sum_{k=1}^{i-1} \left( \lambda_{\Phi_i^*}^{\mu_{k1}}(\mu_{k1}) + \lambda_{\Phi_i^*}^{\mu_{k2}}(\mu_{k2}) \right)$$
(114)

By convenience, let  $\gamma_{e_i}^{e_i} = \text{Id.}$  Define

$$\Pi_{i}(s) = \lambda_{\Phi_{i}^{*}}^{e_{0}} \circ \underline{\alpha}_{0}^{-1} \circ \left(\gamma_{e_{i}}^{e_{0}}\right)^{-1} \circ \alpha_{V}(s) + \sum_{k=1}^{i+1} \lambda_{\Phi_{i}^{*}}^{e_{k}} \circ \alpha_{V}^{-1} \circ \left(\gamma_{e_{i}}^{e_{k}}\right)^{-1} \circ \alpha_{V}(s) + \sum_{k=1}^{i-1} \lambda_{\Phi_{i}^{*}}^{\mu_{k2}} \circ \left(\gamma_{e_{i}}^{\mu_{k2}}\right)^{-1} \circ \alpha_{V}(s) + \sum_{k=1}^{i-1} \lambda_{\Phi_{i}^{*}}^{\mu_{k2}} \circ \left(\gamma_{e_{i}}^{\mu_{k2}}\right)^{-1} \circ \alpha_{V}(s) + \frac{\iota_{i}}{2}s$$

$$(115)$$

for  $s \in \mathbb{R}_+$ . Then,  $\Pi_i \in \mathscr{K}_{\infty}$ .

From Lemma 1 in [41], for any  $0 < c_{i1}, c_{i2} < 1, \varepsilon_i > 0$ , one can find a  $\nu_i : \mathbb{R}_+ \to \mathbb{R}_+$  positive, nondecreasing and continuously differentiable on  $(0, \infty)$  and satisfying

 $(1 - c_{i2})(1 - c_{i1})v_i ((1 - c_{i1})s) s \ge \Pi_i(s)$ (116)

for  $s \ge \sqrt{2\varepsilon_i}$ . With the  $v_i$  satisfying (116), define  $\kappa_i(r) = -v_i(|r|)r$  for  $r \in \mathbb{R}$ . Noticing that  $\lim_{t\to 0^+} \frac{d\kappa_i(r)}{dr} = \lim_{t\to 0^-} \frac{d\kappa_i(r)}{dr}$ ,  $\kappa_i$  is continuously differentiable, odd, strictly decreasing and radially unbounded.

Recall  $V_k(e_k) = \alpha_V(|e_k|) = \frac{1}{2}e_k^2$  for k = 1, ..., n. We use  $V_k$  instead of  $V_k(e_k)$  for k = 1, ..., n. Consider the case of

$$V_{i} \geq \max_{k=1,\dots,i-1} \left\{ \begin{array}{l} \gamma_{e_{i}}^{e_{0}}(V_{0}), \gamma_{e_{i}}^{e_{k}}(V_{k}), \gamma_{e_{i}}^{e_{i+1}}(V_{i+1}), \\ \gamma_{e_{i}}^{\mu_{k1}}(\mu_{k1}), \gamma_{e_{i}}^{\mu_{k2}}(\mu_{k2}), \gamma_{e_{i}}^{\mu_{i1}}(\mu_{i1}), \gamma_{e_{i}}^{\mu_{i2}}(\mu_{i2}), \varepsilon_{i} \end{array} \right\}.$$
(117)

In this case, we have

$$\Pi_{i}(|e_{i}|) - \frac{\iota_{i}}{2}|e_{i}| \ge \phi_{i}^{*}$$
(118)

for all  $\phi_i^* \in \Phi_i^*(e_{i+1}, \bar{x}_i, \bar{\mu}_{(i-1)1}, \bar{\mu}_{(i-1)2}, z)$ . And it also holds that

$$\mu_{i1} \le c_{i1} |e_i|; \quad \mu_{i2} \le c_{i2} \bar{\kappa}_i \left( (1 - c_{i1}) |e_i| \right) |e_i|; \quad |e_i| \ge \sqrt{2\varepsilon_i}. \tag{119}$$

When  $e_i \neq 0$ , with  $0 < c_{i1} < 1$ , for  $\zeta_{i-1} \in S_{i-1}$  and  $|b_{i1}| \le \max\{c_{i1}|e_i|, \mu_{i1}\} = c_{i1}|e_i|$ , we have

$$|x_i - \varsigma_{i-1} + b_{i1}| \ge (1 - c_{i1})|e_i| \tag{120}$$

$$\operatorname{sgn}(x_i - \zeta_{i-1} + b_{i1}) = \operatorname{sgn}(e_i)$$
(121)

and thus

$$\nu_i(|x_i - \varsigma_{i-1} + b_{i1}|)|x_i - \varsigma_{i-1} + b_{i1}| \ge (1 - c_{i1})\nu_i((1 - c_{i1})|e_i|)|e_i|.$$
(122)

In the case of (117), for any  $\phi_i^* \in \Phi_i^*(e_{i+1}, \bar{x}_i, \bar{\mu}_{(i-1)1}, \bar{\mu}_{(i-1)2}, z)$ , with  $\varsigma_{i-1} \in S_{i-1}, |b_{i1}| \le \max\{c_{i1}|e_i|, \mu_{i1}\}$  and  $|b_{i2}| \le \mu_{i2}$ , using (118)–(122), we have

$$\nabla V_{i} \left( \kappa_{i} (x_{i} - \varsigma_{i-1} + b_{i1}) + b_{i2} + \phi_{i}^{*} \right)$$

$$= e_{i} \left( -v_{i} (|x_{i} - \varsigma_{i-1} + b_{i1}|)(x_{i} - \varsigma_{i-1} + b_{i1}) + b_{i2} + \phi_{i}^{*} \right)$$

$$\leq -v_{i} (|x_{i} - \varsigma_{i-1} + b_{i1}|)|x_{i} - \varsigma_{i-1} + b_{i1}||e_{i}| + |b_{i2}||e_{i}| + |\phi_{i}^{*}||e_{i}|$$

$$\leq -(1 - c_{i2})(1 - c_{i1})v_{i}((1 - c_{i1})|e_{i}|)|e_{i}|^{2} + \Pi_{i}(|e_{i}|)|e_{i}| - \frac{\iota_{i}}{2}|e_{i}|^{2}$$

$$\leq -\frac{\iota_{i}}{2}|e_{i}|^{2} = -\iota_{i}\alpha_{V}(|e_{i}|)$$
(123)

which implies (43). This ends the proof.

## C Proof of Lemma 3

For convenience of notations, define  $v_n = u$ . Following the discussions in Remark 9, we have  $v_0 \in S_0(\bar{x}_0, \bar{\mu}_{01}, \bar{\mu}_{02})$ . Suppose that  $v_{i-1} \in S_{i-1}(\bar{x}_{i-1}, \bar{\mu}_{(i-1)1}, \bar{\mu}_{(i-1)2})$ . We will find  $\zeta_{i-1}$ ,  $b_{i1}$  and  $b_{i2}$  satisfying  $\zeta_{i-1} \in S_{i-1}(\bar{x}_{i-1}, \bar{\mu}_{(i-1)1}, \bar{\mu}_{(i-1)2})$ ,  $|b_{i1}| \le \max\{c_{i1}|e_i|, \mu_{i1}\}$  and  $|b_{i2}| \le \mu_{i2}$ , respectively, such that

$$v_i = \kappa_i (x_i - \varsigma_{i-1} + b_{i1}) + b_{i2} \in S_i(\bar{x}_i, \bar{\mu}_{i1}, \bar{\mu}_{i2}).$$
(124)

By applying this reasoning repeatedly, property (50) can be proved. We consider only the case of  $e_i \ge 0$ . The proof for the case of  $e_i < 0$  is similar. We study the following cases (A) and (B).

(A)  $|\kappa_i(q_{i1}(x_i - v_{i-1}, \mu_{i1}))| \le M_{i2}\mu_{i2}$ .

With Assumption 4 satisfied, one can find a  $|b_{i2}| \le \mu_{i2}$  such that

$$q_{i2}(\kappa_i(q_{i1}(x_i - v_{i-1}, \mu_{i1})), \mu_{i2}) = \kappa_i(q_{i1}(x_i - v_{i-1}, \mu_{i1})) + b_{i2}.$$
 (125)

 $(A1) |x_i - v_{i-1}| \le M_{i1} \mu_{i1}.$ 

In this case, Assumption 4 implies that there exists a  $|b_{i1}| \le \mu_{i1}$  such that

$$q_{i1}(x_i - v_{i-1}, \mu_{i1}) = x_i - v_{i-1} + b_{i1}$$
(126)

Choose  $\zeta_{i-1} = v_{i-1}$ . Then,  $\zeta_{i-1} \in S_{i-1}(\bar{x}_{i-1}, \bar{\mu}_{(i-1)1}, \bar{\mu}_{(i-1)2})$  and

$$q_{i1}(x_i - v_{i-1}, \mu_{i1}) = x_i - \varsigma_{i-1} + b_{i1}.$$
(127)

 $(A2) |x_i - v_{i-1}| > M_{i1}\mu_{i1}.$ 

In this case, Assumption 4 implies that there exists a  $|b_{i1}| \le \mu_{i1}$  such that

$$q_{i1}(x_i - v_{i-1}, \mu_{i1}) = \operatorname{sgn}(x_i - v_{i-1})M_{i1}\mu_{i1} + b_{i1}$$
(128)

We study the following two cases.

 $-e_i > 0$ . Recall (37) and (48). In this case, we have  $x_i > v_{i-1}$  and

$$x_i - v_{i-1} > M_{i1}\mu_{i1} \ge e_i = x_i - \max S_{i-1}(\bar{x}_{i-1}, \bar{\mu}_{(i-1)1}, \bar{\mu}_{(i-1)2}).$$
 (129)

One can find a  $\zeta_{i-1} \in [v_{i-1}, \max S_{i-1}(\bar{x}_{i-1}, \bar{\mu}_{(i-1)1}, \bar{\mu}_{(i-1)2})]$  such that  $x_i - \zeta_{i-1} = M_{i1}\mu_{i1}$  and thus

$$q_{i1}(x_i - v_{i-1}, \mu_{i1}) = x_i - \varsigma_{i-1} + b_{i1}.$$
(130)

-  $e_i = 0$ . In this case, using (37), we have  $x_i \in S_{i-1}(\bar{x}_{i-1}, \bar{\mu}_{(i-1)1}, \bar{\mu}_{(i-1)2})$  and can directly find a  $\varsigma_{i-1} \in S_{i-1}(\bar{x}_{i-1}, \bar{\mu}_{(i-1)1}, \bar{\mu}_{(i-1)2})$ , which is closer to  $x_i$  than  $v_{i-1}$  such that  $x_i - \varsigma_{i-1} = \operatorname{sgn}(x_i - v_{i-1})M_{i1}\mu_{i1}$  and thus

$$q_{i1}(x_i - v_{i-1}, \mu_{i1}) = x_i - \varsigma_{i-1} + b_{i1}.$$
(131)

From (127) and (131), in the case of  $|\kappa_i(q_{i1}(x_i - v_{i-1}))| \le M_{i2}\mu_{i2}$ , we can find  $\zeta_{i-1} \in S_{i-1}(\bar{x}_{i-1}), |b_{i1}| \le \mu_{i1}$  and  $|b_{i2}| \le \mu_{i2}$  such that

$$v_i = q_{i2}(\kappa_i(q_{i1}(x_i - v_{i-1}))) = \kappa_i(x_i - \varsigma_{i-1} + b_{i1}) + b_{i2}.$$
 (132)

(B)  $|\kappa_i(q_{i1}(x_i - v_{i-1}, \mu_{i1}))| > M_{i2}\mu_{i2}$ .

Before the discussions, we give the following lemma.

**Lemma 9** Under the conditions of Lemma 3, if  $|\kappa_i(q_{i1}(x_i - v_{i-1}, \mu_{i1}))| > M_{i2}\mu_{i2}$ , then

$$\operatorname{sgn}(x_i - v_{i-1}) = \operatorname{sgn}(q_{i1}(x_i - v_{i-1}, \mu_{i1})).$$
(133)

## The proof of Lemma 9 is in Appendix C.1

Note that  $\kappa_i$  is an odd and strictly decreasing function. We have

$$\operatorname{sgn}(\kappa_i(q_{i1}(x_i - v_{i-1}, \mu_{i1}))) = -\operatorname{sgn}(q_{i1}(x_i - v_{i-1}, \mu_{i1})) = -\operatorname{sgn}(x_i - v_{i-1}).$$
(134)

Under Assumption 4, using Lemma 9 one can find a  $|b_{i2}| \le \mu_{i2}$  such that

$$q_{i2}(\kappa_i(q_{i1}(x_i - v_{i-1}, \mu_{i1})), \mu_{i2})$$
  
= sgn( $\kappa_i(q_{i1}(x_i - v_{i-1}, \mu_{i1})))M_{i2}\mu_{i2} + b_{i2}$   
= -sgn( $x_i - v_{i-1})M_{i2}\mu_{i2} + b_{i2}.$  (135)

 $(B1) e_i > 0.$ 

In this case, using (37), we have  $x_i > v_{i-1}$  and thus

$$\kappa_i(q_{i1}(x_i - v_{i-1}, \mu_{i1})) < 0, \tag{136}$$

$$q_{i2}(\kappa_i(q_{i1}(x_i - v_{i-1}, \mu_{i1}))) = -M_{i2}\mu_{i2} + b_{i2}.$$
(137)

With  $|\kappa_i(q_{i1}(x_i - v_{i-1}, \mu_{i1}))| > M_{i2}\mu_{i2}$ , property (136) implies

$$\kappa_i(q_{i1}(x_i - v_{i-1}, \mu_{i1})) < -M_{i2}\mu_{i2}.$$
(138)

Consider the following two cases.

−  $x_i - v_{i-1} \le M_{i1} \mu_{i1}$ . In this case, under Assumption 4, one can find a  $|b'_{i1}| \le \mu_{i1}$  such that

$$\kappa_i(x_i - v_{i-1} + b'_{i1}) = \kappa_i(q_{i1}(x_i - v_{i-1}, \mu_{i1})) < -M_{i2}\mu_{i2}.$$
 (139)

-  $x_1 - v_{i-1} > M_{i1}\mu_{i1}$ . In this case, under Assumption 4, one can find a  $|b'_{i1}| \le \mu_{i1}$  such that

$$\kappa_i(M_{i1}\mu_{i1} + b'_{i1}) = \kappa_i(q_{i1}(x_i - v_{i-1}, \mu_{i1})) < -M_{i2}\mu_{i2}.$$
(140)

Using the strictly decreasing property of  $\kappa_i$ , we have

$$\kappa_i(x_i - v_{i-1} + b'_{i1}) < \kappa_i(M_{i1}\mu_{i1} + b'_{i1}) < -M_{i2}\mu_{i2}.$$
(141)

Thus, in both the cases above, one can find a  $|b'_{i1}| \le \mu_{i1}$  such that

$$\kappa_i(x_i - v_{i-1} + b'_{i1}) < \kappa_i(M_{i1}\mu_{i1} + b'_{i1}) < -M_{i2}\mu_{i2}.$$
(142)

From (49), we have

$$\bar{\kappa}_i((1-c_{i1})|e_i|) < M_{i2}\mu_{i2}. \tag{143}$$

Deringer

From the definition of  $e_i$ , using  $e_i > 0$ , we have

$$\kappa_i(x_i - \max S_{i-1}(\bar{x}_{i-1}, \bar{\mu}_{(i-1)1}, \bar{\mu}_{(i-1)2}) - c_{i1}e_i) > -M_{i2}\mu_{i2}.$$
 (144)

From (142) and (144), using the continuity of  $\kappa_i$ , one can find a  $\varsigma_{i-1} \in [v_{i-1}, \max S_{i-1}(\bar{x}_{i-1}, \bar{\mu}_{(i-1)1}, \bar{\mu}_{(i-1)2})]$  and a  $b_{i1} \in [-c_{i1}e_i, b'_{i1}]$  such that

$$\kappa_i(x_i - \varsigma_{i-1} + b_{i1}) = -M_{i2}\mu_{i2}.$$
(145)

Recall (137). We have

$$v_i = q_{i2}(\kappa_i(q_{i1}(x_i - v_{i-1}, \mu_{i1}))) = \kappa_i(x_i - \varsigma_{i-1} + b_{i1}) + b_{i2}.$$
 (146)

 $(B2) e_i = 0.$ 

In this case,  $x_i \in S_{i-1}(\bar{x}_{i-1}, \bar{\mu}_{(i-1)1}, \bar{\mu}_{(i-1)2})$ . From Lemma 9, we have  $x_i - v_{i-1} \neq 0$ . Consider the following two cases.

-  $|x_i - v_{i-1}| \le M_{i1}\mu_{i1}$ . In this case, define  $\varsigma'_{i-1} = v_{i-1}$ . With Assumption 4, one can find a  $|b'_{i1}| \le \mu_{i1}$  such that

$$\kappa_{i}(x_{i} - \varsigma_{i-1}' + b_{i1}') = \kappa_{i}(q_{i1}(x_{i} - v_{i-1}, \mu_{i1})) \begin{cases} > M_{i2}\mu_{i2}, & \text{if } x_{i} < \varsigma_{i-1}' \\ < -M_{i2}\mu_{i2}, & \text{if } x_{i} > \varsigma_{i-1}'. \end{cases}$$
(147)

We used  $|\kappa_i(q_{i1}(x_i - v_{i-1}, \mu_{i1}))| > M_{i2}\mu_{i2}$  and (134) for the last part of (147). -  $|x_i - v_{i-1}| > M_{i1}\mu_{i1}$ . In this case, under Assumption 4, one can find a  $|b'_{i1}| \le \mu_{i1}$  such that

$$\kappa_{i}(\operatorname{sgn}(x_{i} - v_{i-1})M_{i1}\mu_{i1} + b'_{i1}) = \kappa_{i}(q_{i1}(x_{i} - v_{i-1}, \mu_{i1})) \begin{cases} > M_{i2}\mu_{i2}, & \text{if } x_{i} < v_{i-1} \\ < -M_{i2}\mu_{i2}, & \text{if } x_{i} > v_{i-1}. \end{cases}$$
(148)

In the case of  $|x_i - v_{i-1}| > M_{i1}\mu_{i1}$ , one can find a  $\varsigma'_{i-1} \in [x_i, v_{i-1}]$  satisfying  $\operatorname{sgn}(x_i - \varsigma'_{i-1}) = \operatorname{sgn}(x_i - v_{i-1})$  and  $\operatorname{sgn}(x_i - v_{i-1})M_{i1}\mu_{i1} = x_i - \varsigma'_{i-1}$ . In this way, we achieve

$$\kappa_i(x_i - \varsigma'_{i-1} + b'_{i1}) \begin{cases} > M_{i2}\mu_{i2}, & \text{if } x_i < \varsigma'_{i-1} \\ < -M_{i2}\mu_{i2}, & \text{if } x_i > \varsigma'_{i-1}. \end{cases}$$
(149)

Note that  $\kappa_i(x_i - x_i + 0) = \kappa_i(0) = 0$ . By using the continuity of  $\kappa_i$ , one can find a  $\varsigma_{i-1} \in [x_i, \varsigma'_{i-1}]$  and a  $b_{i1} \in [0, b'_{i1}]$  such that

$$sgn(x_i - \zeta_{i-1}) = sgn(x_i - \zeta'_{i-1}) = sgn(x_i - v_{i-1})$$
(150)

$$\kappa_i(x_i - \varsigma_{i-1} + b_{i1}) = -\operatorname{sgn}(x_i - v_{i-1})M_{i2}\mu_{i2}.$$
(151)

Clearly,  $\zeta_{i-1} \in S_{i-1}(\bar{x}_{i-1}, \bar{\mu}_{(i-1)1}, \bar{\mu}_{(i-1)2})$  and  $|b_{i1}| \leq \mu_{i1}$ . Recall (135). We have

$$v_i = q_{i2}(\kappa_i(q_{i1}(x_i - v_{i-1}, \mu_{i1}))) = \kappa_i(x_i - \varsigma_{i-1} + b_{i1}) + b_{i2}.$$
 (152)

Considering both cases (A) and (B), the proof of Lemma 3 is concluded.

# C.1 Proof of Lemma 9

Consider the following two cases.

-  $|x_i - v_{i-1}| > M_{i1}\mu_{i1}$ . In this case, under Assumption 4, one can find a  $|b_{i1}| \le \mu_{i1}$  such that

$$q_{i1}(x_i - v_{i-1}, \mu_{i1}) = \operatorname{sgn}(x_i - v_{i-1})M_{i1}\mu_{i1} + b_{i1}$$
(153)

Note that  $M_{i1} > 2$ . Thus,

$$sgn(x_i - v_{i-1}) = sgn(q_{i1}(x_i - v_{i-1}, \mu_{i1})).$$
(154)

-  $|x_i - v_{i-1}| \le M_{i1}\mu_{i1}$ . In this case, under Assumption 4, one can find a  $|b_{i1}| \le \mu_{i1}$  such that

$$q_{i1}(x_i - v_{i-1}, \mu_{i1}) = x_i - v_{i-1} + b_{i1}.$$
(155)

Condition  $|\kappa_i(q_{i1}(x_i - v_{i-1}, \mu_{i1}))| > M_{i2}\mu_{i2}$  implies  $q_{i1}(x_i - v_{i-1}, \mu_{i1}) \neq 0$ . If  $sgn(x_i - v_{i-1}) \neq sgn(q_{i1}(x_i - v_{i-1}, \mu_{i1}))$ , then  $sgn(b_{i1}) = sgn(q_{i1}(x_i - v_{i-1}, \mu_{i1}))$  and  $|b_{i1}| > |x_i - v_{i-1}|$ . Thus,  $|x_i - v_{i-1} + b_{i1}| \leq |b_{i1}| \leq \mu_{i1}$ . Note that  $\frac{1}{M_{i1}} < c_{i1} < 0.5$ . Then, we can derive

$$\begin{aligned} |\kappa_i(q_{i1}(x_i - v_{i-1}, \mu_{i1}))| &\leq \bar{\kappa}_i(\mu_{i1}) \leq \bar{\kappa}_i \left(\frac{1 - c_{i1}}{c_{i1}}\mu_{i1}\right) < \bar{\kappa}_i((1 - c_{i1})M_{i1}\mu_{i1}) \\ &= M_{i2}\mu_{i2} \end{aligned}$$
(156)

which leads to a contradiction with  $|\kappa_i(q_{i1}(x_i - v_{i-1}))| > M_{i2}\mu_{i2}$ . We used (60) for the last equality in (156). This ends the proof of Lemma 9.

# D Proof of Lemma 6

Recall that if  $\chi_1, \chi_2 \in \mathscr{K}_{\infty}$  satisfies  $\chi_1(s) > \chi_2(s)$  for  $s \in \mathbb{R}_+$ , then  $(\mathrm{Id} - \tilde{\chi}) \circ \chi_1(s) \ge \chi_2(s)$  for  $s \in \mathbb{R}_+$  with  $\tilde{\chi} := \mathrm{Id} - \chi_2 \circ \chi_1^{-1}$  being continuous and positive definite.

For each i = 1, ..., n, with (47) satisfied, we can find a continuous and positive definite  $\rho_i$  such that

$$\sigma_i \circ \alpha_V \left(\frac{1}{c_{i1}}s\right) \le (\mathrm{Id} - \rho_i) \circ \sigma_i \circ \alpha_V(M_{i1}s) \tag{157}$$

$$\sigma_i \circ \alpha_V \left( \frac{1}{1 - c_{i1}} \bar{\kappa}_i^{-1} \left( \frac{1}{c_{i2}} s \right) \right) \le (\mathrm{Id} - \rho_i) \circ \sigma_i \circ \alpha_V \left( \frac{1}{1 - c_{i1}} \bar{\kappa}_i^{-1} \left( M_{i2} s \right) \right)$$
(158)

for all  $s \in \mathbb{R}_+$ . Define  $\rho(s) = \min_{i=1,...,n} \{\rho_i(s)\}$  for  $s \in \mathbb{R}_+$ . Then,  $\rho$  is continuous and positive definite. Using (58), (59), (64) and (69), we have

$$B_2(\bar{\mu}_{n1}(t), \bar{\mu}_{n2}(t)) \le (\mathrm{Id} - \rho)(\Theta(t))$$
 (159)

for any  $t \in \mathbb{R}_+$ .

Note that the zooming variables  $\mu_{n1}(t)$  and  $\mu_{n2}(t)$  are constant on  $[t_k, t_{k+1})$ , that is,  $\mu_{n1}(t) = \mu_{n1}(t_k)$  and  $\mu_{n2}(t) = \mu_{n2}(t_k)$  for  $t \in [t_k, t_{k+1})$ .

Suppose (79) holds. We study the following two cases:

- (a)  $V(e(X(t_{k+1}), \bar{\mu}_{n1}(t_k), \bar{\mu}_{n2}(t_k))) < \max\{(\mathrm{Id} \rho)(\Theta(t_k)), \theta_0\}.$
- (b)  $V(e(X(t_{k+1}), \bar{\mu}_{n1}(t_k), \bar{\mu}_{n2}(t_k))) \ge \max\{(\mathrm{Id} \rho)(\Theta(t_k)), \theta_0\}$ . In this case, from (63), (79) and (159), it follows that  $V(e(X(t), \bar{\mu}_{n1}(t), \bar{\mu}_{n2}(t)))$  is strictly decreasing for  $t \in [t_k, t_{k+1})$  and

$$\max\{(\mathrm{Id} - \rho)(\Theta(t_k)), \theta_0\} \le V(e(X(t), \bar{\mu}_{n1}(t), \bar{\mu}_{n2}(t))) \le \Theta(t_k)$$
(160)

for all  $t \in [t_k, t_{k+1})$ . Using (63), we have

$$V(e(X(t_{k+1}), \bar{\mu}_{n1}(t_k), \bar{\mu}_{n2}(t_k)))$$

$$\leq V(e(X(t_k), \bar{\mu}_{n1}(t_k), \bar{\mu}_{n2}(t_k))) - \int_{t_k}^{t_{k+1}} \alpha(V(e(X(\tau), \bar{\mu}_{n1}(\tau), \bar{\mu}_{n2}(\tau)))) d\tau$$

$$\leq \Theta(t_k) - t_d \cdot \min_{\substack{\max\{(\mathrm{Id}-\rho)(\Theta(t_k)), \theta_0\} \leq v \leq \Theta(t_k)}} \alpha(v)$$

$$\leq \Theta(t_k) - t_d \cdot \min_{\substack{(\mathrm{Id}-\rho)(\Theta(t_k)), \delta v \leq v \leq \Theta(t_k)}} \alpha(v)$$
(161)

where  $t_d = t_{k+1} - t_k$ . Define  $\rho'(s) = t_d \cdot \min_{(\mathrm{Id} - \rho)(s) \le v \le s} \alpha(v)$  for  $s \in \mathbb{R}_+$ . Then, it can be directly verified that  $\rho'$  is continuous and positive definite and that

$$V(e(X(t_{k+1}), \bar{\mu}_{n1}(t_k), \bar{\mu}_{n2}(t_k))) \le (\mathrm{Id} - \rho')(\Theta(t_k)).$$
(162)

Lemma 6 is proved by finding a continuous and positive definite function  $\bar{\rho}$  such that  $(\mathrm{Id} - \bar{\rho}) \in \mathscr{K}_{\infty}$  and  $(\mathrm{Id} - \bar{\rho})(s) \ge \max\{(\mathrm{Id} - \rho)(s), (\mathrm{Id} - \rho')(s)\}$  for  $s \in \mathbb{R}_+$ .

# References

- Brockett RW, Liberzon D (2000) Quantized feedback stabilization of linear systems. IEEE Trans Autom Control 45:1279–1289
- 2. Miller RK, Mousa MS, Michel AN (1988) Quantization and overflow effects in digital implementation of linear dynamic controllers. IEEE Trans Autom Control 33:698–704
- Delchamps DF (1990) Stabilizing a linear system with quantized state feedback. IEEE Trans Autom Control 35:916–924
- Fu M, Xie L (2005) The sector bound approach to quantized feedback control. IEEE Trans Autom Control 50:1698–1711
- Elia N, Mitter SK (2001) Stabilization of linear systems with limited information. IEEE Trans Autom Control 46:1384–1400
- Ceragioli F, De Persis C (2007) Discontinuous stabilization of nonlinear systems: quantized and switching controls. Syst Control Lett 56:461–473
- 7. De Persis C (2009) Robust stabilization of nonlinear systems by quantized and ternary control. Syst Control Lett 58:602–608
- Liberzon D, Hespanha JP (2005) Stabilization of nonlinear systems with limited information feedback. IEEE Trans Autom Control 50:910–915
- Liberzon D, Nešić D (2007) Input-to-state stabilization of linear systems with quantized state measurements. IEEE Trans Autom Control 52:767–781
- Grüne L (2002) Asymptotic behavior of dynamical and control systems under perturbation and discretization. Lecture Notes in Mathematics, vol 1783. Springer, Berlin
- 11. Liberzon D (2003) Hybrid feedback stabilization of systems with quantized signals. Automatica 39:1543–1554
- 12. Liu T, Jiang ZP, Hill DJ (2012) A sector bound approach to feedback control of nonlinear systems with state quantization. Automatica 48:145–152
- Teel AR, Praly L (1995) Tools for semiglobal stabilization by partial state and output feedback. SIAM J Control Optim 33:1443–1488
- De Persis C (2005) *n*-bit stabilization of *n*-dimensional nonlinear systems in feedforward form. IEEE Trans Autom Control 50:299–311
- 15. Liberzon D (2003) Switching in systems and control. Birkhäuser, Boston
- Ledyaev YS, Sontag ED (1999) A Lyapunov characterization of robust stabilization. Nonlinear Anal 37:813–840
- 17. Freeman RA, Kokotović PV (1996) Robust nonlinear control design: state-space and Lyapunov techniques. Birkhäuser, Boston
- Krstić M, Kanellakopoulos I, Kokotović PV (1995) Nonlinear and adaptive control design. Wiley, New York
- Sontag ED (2007) Input to state stability: basic concepts and results. In: Nistri P, Stefani G (eds) Nonlinear and optimal control theory. Springer, Berlin, pp 163–220
- Liberzon D (2006) Quantization, time delays, and nonlinear stabilization. IEEE Trans Autom Control 51:1190–1195
- Liberzon D (2008) Observer-based quantized output feedback control of nonlinear systems. In: Proceedings of the 17th IFAC World Congress, pp 8039–8043
- Jiang ZP, Teel AR, Praly L (1994) Small-gain theorem for ISS systems and applications. Math Control Signals Syst 7:95–120
- Nešić D, Liberzon D (2009) A unified framework for design and analysis of networked and quantized control systems. IEEE Trans Autom Control 54:732–747
- Jiang ZP, Wang Y (2008) A generalization of the nonlinear small-gain theorem for large-scale complex systems. In: Proceedings of the 7th World congress on intelligent control and automation, pp 1188–1193
- Liu T, Hill DJ, Jiang ZP (2011) Lyapunov formulation of ISS cyclic-small-gain in continuous-time dynamical networks. Automatica 47:2088–2093
- 26. Liu T, Jiang ZP, Hill DJ (2012) Small-gain based output-feedback controller design for a class of nonlinear systems with actuator dynamic quantization. IEEE Trans Autom Control (in press)
- 27. Jiang ZP, Mareels IMY, Wang Y (1996) A Lyapunov formulation of the nonlinear small-gain theorem for interconnected systems. Automatica 32:1211–1214
- 28. Teel AR Input-to-state stability and the nonlinear small-gain theorem. Private communication

- Potrykus HG, Allgöwer F, Qin SJ (2003) The character of an idempotent-analytic nonlinear small gain theorem. In: Positive systems. Lecture notes in control and information science. Springer, Berlin, pp 361–368
- Dashkovskiy S, Rüffer BS, Wirth FR (2007) An ISS small-gain theorem for general networks. Math Control Signals Syst 19:93–122
- Dashkovskiy S, Rüffer BS, Wirth FR (2010) Small gain theorems for large scale systems and construction of ISS Lyapunov functions. SIAM J Control Optim 48:4089–4118
- 32. Rüffer BS (2007) Monotone systems, graphs, and stability of large-scale interconnected systems. Ph.D. thesis, University of Bremen, Germany. http://nbn-resolving.de/urn:nbn:de:gbv:46-diss000109058
- Heemels WPMH, Weiland S (2007) Input-to-state stability and interconnections of discontinuous dynamical systems. Tech. Rep. DCT report 2007-104, Eindhoven University of Technology. http:// repository.tue.nl/656556
- Heemels WPMH, Weiland S (2008) Input-to-state stability and interconnections of discontinuous dynamical systems. Automatica 44:3079–3086
- 35. Filippov AF (1988) Differential equations with discontinuous righthand sides. Kluwer Academic Publishers, Dordrecht
- 36. Federer H (1969) Geometric measure theory. Springer, New York
- Hespanha JP, Liberzon D, Angeli D, Sontag ED (2005) Nonlinear norm-observability notions and stability of switched systems. IEEE Trans Autom Control 50:154–168
- Praly L, Jiang ZP (1993) Stabilization by output feedback for systems with ISS inverse dynamics. Syst Control Lett 21:19–33
- Jiang ZP (1999) A combined backstepping and small-gain approach to adaptive output-feedback control. Automatica 35:1131–1139
- Liu T, Jiang ZP, Hill DJ (2011) Quantized output-feedback control of nonlinear systems: a cyclicsmall-gain approach. In: Proceedings of the 30th Chinese Control Conference, pp 487–492
- Jiang ZP, Mareels IMY (1997) A small-gain control method for nonlinear cascade systems with dynamic uncertainties. IEEE Trans Autom Control 42:292–308
- Sharon Y, Liberzon D (2012) Input to state stabilizing controller for systems with coarse quantization. https://netfiles.uiuc.edu/liberzon/www/research/yoav-coarse-quant-iss.pdf (in press)
- Praly L, Wang Y (1996) Stabilization in spite of matched unmodeled dynamics and an equivalent definition of input-to-state stability. Math Control Signals Syst 9:1–33
- Liu T, Jiang ZP, Hill DJ (2011) Quantized output-feedback control of nonlinear systems: a cyclic-smallgain approach. Tech. rep., The Australian National University. http://users.cecs.anu.edu.au/tengfei/TR. pdf
- 45. Karafyllis I, Jiang ZP (2011) Stability and stabilization of nonlinear systems. Springer, Berlin
- 46. Yakubovich VA, Leonov GA, Gelig AK (2004) Stability of stationary sets in control systems with discontinuous nonlinearities. World Scientific Press, Singapore